### Bank of England

# The effects of subsidised flood insurance on real estate markets

#### **Staff Working Paper No. 995**

October 2024

Nicola Garbarino, Benjamin Guin and Jonathan Lee

This is an updated version of the Staff Working Paper originally published on 30 September 2022

Staff Working Papers describe research in progress by the author(s) and are published to elicit comments and to further debate. Any views expressed are solely those of the author(s) and so cannot be taken to represent those of the Bank of England or any of its committees, or to state Bank of England policy.



## **Bank of England**

#### Staff Working Paper No. 995

# The effects of subsidised flood insurance on real estate markets

Nicola Garbarino,<sup>(1)</sup> Benjamin Guin<sup>(2)</sup> and Jonathan Lee<sup>(3)</sup>

#### Abstract

Subsidised insurance against extreme weather events improves affordability among households in at-risk areas but it can weaken the risk signal via property prices. Leveraging a granular data set of all property transactions and flood events in England, we study the effects of a reinsurance scheme which lowers insurance premiums for at-risk properties. We document that the introduction of this scheme increases prices and transaction volumes of flood-prone properties. This fully offsets the negative direct effects of flooding on property prices, with high-income areas and high-value properties benefiting relatively more. Our findings speak to the debate on climate adaptation policies and their consequences for wealth distribution.

Key words: House prices, flood risk, flood insurance, climate risks.

JEL classification: G21, Q54.

- (1) Ludwig Maximilian University of Munich and Ifo Institute for Economic Research. Email: garbarino@ifo.de
- (2) Bank of England. Email: benjamin.guin@bankofengland.co.uk
- (3) University of Glasgow. Email: jonathan.lee.2@glasgow.ac.uk

For comments and discussions we would like to thank Allan Beltrán, Laura Ballester, Erik Benrud, Martin Brown, Celso Brunetti, Stefan Claus, Piotr Danisewicz, Julian Gray, Eleanor Hammerton, Sinem Hacioglu Hoke, Zhongchen Hu, Huyen Nguyen, Jose A Poncela, Klaus Schaeck, Paolo Siciliani, Mike Steel, Swenja Surminski, an anonymous referee of the Staff Working Paper Series and staff of Flood Re, the Environment Agency and Defra. We also thank seminar participants at the Bank of England, the Green Seminar at the European Central Bank, the European System of Central Banks, the 2022 Meeting of the EFMA and the 27th Conference of the EAERE, the 2023 World Finance Conference and the 3rd Annual Conference of the ASFAAG. We are grateful to Zahid Amadxarif for his research assistance. Any views expressed are solely those of the authors. They should not be taken to represent those of the Bank of England or as a statement of Bank of England policy. This paper should not be reported as representing the views of members of the Monetary Policy Committee, Prudential Regulation Committee or Financial Policy Committee. Nicola Garbarino's contribution was partially supported by the German Ministry of Education and Research (BMBF) under the project 'Klimawandelanpassung auf regionaler Ebene: ansteigende Starkregenrisiken am Beispiel des bayerischen Oberlandes' (KARE; grant number 01LR2006A). This version was submitted to the Journal of Risk and Insurance. The Bank's working paper series can be found at www.bankofengland.co.uk/working-paper/staff-working-papers

Bank of England, Threadneedle Street, London, EC2R 8AH Email: enquiries@bankofengland.co.uk

©2024 Bank of England ISSN 1749-9135 (on-line)

#### 1 Introduction

Housing is vulnerable to extreme weather events related to climate change (Gourevitch et al., 2023).<sup>1</sup> At the same time, it is one of the major vehicles of household wealth accumulation (Bhatia, 1987; Benjamin et al., 2004; Bin et al., 2008; Bach et al., 2020) and one of the major types of collateral in the financial system (Chaney et al., 2012; Ramcharan, 2020). The literature on the effect of extreme weather events on property values is extensive (e.g. Hallstrom and Smith (2005); Beltrán et al. (2018)). Less is known about the role in property markets of public policies against extreme weather events.

To address this gap, we exploit a novel empirical setting, the introduction of a UK public scheme, Flood Re, which provides insurers with an option to re-insure the flood risk element of their policies at a pre-determined (not risk-based) price. Flood Re was introduced in April 2016 with the aim of improving the affordability of flood insurance in flood-prone areas (Flood Re, 2016), and it is estimated to halve the average annual insurance premium of flooded properties.<sup>2</sup> The scheme is financed by a levy on all UK home insurance policies, resulting in a small increase in premiums in areas not exposed to flood risk.<sup>3</sup>

Beyond the introduction of Flood Re, the UK residential real estate market offers an appealing setting to study the introduction of a public flood reinsurance scheme for several reasons. First, homeownership rates in the UK are high (about two-thirds of households own a property, Campbell (2013)). Second, mortgage lenders require building insurance to extend a loan, and as a result take-up rates of home insurance, which also covers flood risk, are high, reaching over 95% in England, also in risky areas (Surminski, 2018; Defra, 2018). The high take-up rate allows us to estimate the effect of Flood Re on property prices without explicitly looking at the level of insurance coverage. Third, information on flood risk is publicly available to all participants in the real estate market. The UK Environment Agency

<sup>&</sup>lt;sup>1</sup>For the United Kingdom, the Environment Agency estimates that one in every six properties, in total 5.2 millions properties, across England are at risk of flooding. In the U.S., the National Oceanic and Atmospheric Administration estimates that \$106 billion worth of coastal property will be below sea level by 2050. <sup>2</sup>From around £650 to less than £325, based on survey data from DEFRA (2013).

From around 2000 to less than 2525, based on survey data from DEFRA (20.

<sup>&</sup>lt;sup>3</sup>The increase is estimated at around £10.5 annually (Surminski, 2018).

has been publishing granular flood maps since 2004 (Belanger and Bourdeau-Brien, 2018).

In this setting, we study the effect of Flood Re on property prices. To isolate the effect of the policy from other confounding factors driving property prices we leverage a comprehensive data set of all property transactions (including the full address) in England.<sup>4</sup> To control for local factors and unobservable but time-invariant property characteristics, we compare price changes for individual properties in a small area. The data set also allows us to control for observable property characteristics, such as property type, on transaction prices. To measure flood risk we use both flood risk maps as well as maps with the outlines of actual flood events, published by the Environment Agency (EA). We focus on properties built before 2002, when tighter building standards were introduced for new developments in flood-prone areas.

We find that flood events reduce property values before the introduction of Flood Re (by 1.6% in our preferred specification). After the introduction of Flood Re, we do not observe a reduction in the values of flooded properties. On average, the introduction of Flood Re increases the value of flooded properties by  $\pounds 4,083.^5$ 

The high insurance take-up rate means that properties damaged by floods are likely to be repaired before being sold, mitigating concerns about differences in the physical state of buildings. We further address this concern by using flood risk maps, which provide an ex-ante measure of flood risk. Compared to no-risk properties, at-risk properties are sold at discount before the introduction of Flood Re. The discount (before Flood Re) on at-risk properties (0.4%) is lower than on flooded properties (1.6%). This result aligns with recent findings by Niu et al. (2023), which indicate that ex-ante flood risk has a smaller impact on property prices due to its lower salience. The discount vanishes after introduction of Flood Re, implying that our findings are related to the expectation of future flood risk.

<sup>&</sup>lt;sup>4</sup>See Lustig and Van Nieuwerburgh (2005) and Piazzesi et al. (2007) for other drivers of property prices.

<sup>&</sup>lt;sup>5</sup>The average property price is £226,840. The calculation is based on the estimates of our preferred specification shown in Column 3 of Table 3: £226,840 × 1.8% = £4,083.

We conduct a set of placebo tests based on a temporary extension of an agreement between the government and insurance providers shortly before the introduction of Flood Re. We do not find any effect of the extension (which did not involve any policy change) on property values. We also conduct simulations by testing the placebo effect of flood events and Flood Re on properties that are not actually flooded. These simulations confirm that our findings are unlikely to be driven by factors other than flood events and Flood Re. Additionally, our results are consistent in a series of robustness test, such as excluding flats from the sample (flats above ground floor are not exposed to direct flood damage) and controlling for the change in hedonic price function after flood (flood events may increase demand for newer properties with stronger flood defences, relative to older properties in the same areas).

The mitigating effects of Flood Re on prices of properties exposed to flood risk is stronger in areas with higher average income, education level and age. The net housing wealth effect (measured as a percentage of average household annual income) is about 14.2% in high-income areas compared to 3.8% in low-income areas. These results highlight possible unintended distributional implications of Flood Re, with benefits concentrated in areas with higher socioeconomic status. We do not find any differences in the effect of Flood Re in terms of beliefs about climate change (proxied by the share of votes to the Green Party).

Finally, we explore the mechanisms through which Flood Re affects property prices. First, we find that Flood Re increases transaction probabilities for properties in at-risk areas. The combination of more transactions and higher prices following Flood Re suggests that buyers increase their demand for properties in at-risk areas (rather than potential sellers reducing their supply). Second, flood events have a larger negative effect on property prices when they hit at-risk areas, which seems consistent with the idea that *ex-post* flood events reinforce existing information about *ex-ante* flood risk. Third, our results suggest that properties built after 2002 do not benefit from the introduction of Flood Re, given their stronger resilience against flood risk.

Our paper contributes to the growing literature examining the linkage between climate risks and government interventions. Several papers consider the effect of public policies, including public information on flood risk (Hu, 2022) and improving flood risk classifications (Mulder, 2021), on flood insurance take up. Wagner (2022) finds that homeowners' willingness to pay for natural disaster insurance is inefficiently low in the absence of government intervention, and flood insurance mandates could increase social welfare. Oh et al. (2022) study the effect of regulated flood insurance prices across U.S. states, and find that insurers overcome pricing frictions by cross-subsidising insurance across states. Our paper shows the effect of a subsidized public flood reinsurance scheme on prices and liquidity in the real-estate market, providing a link to the literature on climate risks and property markets (Bernstein et al., 2019; Baldauf et al., 2020; Murfin and Spiegel, 2020; Nguyen et al., 2022).

We also highlight the distributional impact of flood insurance scheme, contributing to the literature on the distributional impact of environmental policies, e.g., Grainger (2012); Bento et al. (2015); da Silva Freitas et al. (2016); Isen et al. (2017). Studies on the distributional effects of flood insurance schemes focus on the National Flood Insurance Program (NFIP) in the United States. The evidence suggests that the NFIP mitigates the adverse financial impact of flooding better than ex-post disaster assistance (Billings et al., 2022), but also that its net premiums (premiums minus payouts) are regressive, implying that the NFIP disproportionally benefits wealthier segments of population (Bin et al., 2017). These distributional impacts however do not translate to county level, suggesting that subsidized flood insurance does not contribute to regional inequalites (Bin et al., 2012). We study the distributional effect of public flood insurance schemes in a novel setting (the UK Flood Re scheme), in terms of property values rather than insurance premiums, and document that the mitigating effect of a public flood reinsurance scheme on at-risk properties are much stronger in richer areas.

The rest of the paper is structured as follows. Section 2 summarizes the policy background; Section 3 details the data of our analysis; Section 4 presents the empirical strategy; Section 5 discusses the results. We further explore the heterogeneous effects of Flood Re and examine the mechanisms by which Flood Re influences property prices, respectively in Section 6 and 7. Section 8 evaluates housing welath net effects and Section 9 concludes.

#### 2 Policy background

Flood Re was established by the UK government in April 2016 to lower the cost of flood insurance in high-risk areas. The scheme offers reinsurance at a subsidized price linked to the council tax band of the property, rather than its flood risk. Insurance firms can transfer the flood risk component of home insurance policies to Flood Re, which must accept all eligible policies. There is no threshold for claims payout. Properties built after 2009 are excluded to discourage new developments in flood-prone areas.<sup>6</sup> The scheme is financed through a £180 million annual levy on insurers, which is proportional to their share in the home insurance market, along with premium and investment income. Flood Re is scheduled to be phased out in 2039, allowing the flood insurance market to transition to risk-based pricing.

According to Flood Re's 2020 annual report, 80% of households with previous flood claims found quotes that are more than 50% cheaper after the reinsurance scheme started operating (Flood Re, 2020). The average annual insurance premium for flooded properties was roughy halved from around £650 (DEFRA, 2013) to less than £325. Before the introduction of Flood Re, none of the households with prior flood claims received quotes from more than four insurers. However, 94% of them can receive quotes from five or more insurers after the introduction of Flood Re. Since Flood Re's reinsurance nature does not offer insurance policies directly to households, they do not have to be aware of Flood Re to benefit from it. Around half of households in at-risk areas are aware of Flood Re, while only a quarter of households in no-risk areas are familiar with Flood Re (Defra, 2018). We outline the relationship between Flood Re, government and industry in Figure A.1 in our Online Appendix (based on Crick et al. (2018)).

<sup>&</sup>lt;sup>6</sup>Despite of that, a large number of properties are still being built in flood-prone areas, particularly in deprived neighbourhoods (Rözer and Surminski, 2021).

Prior to the introduction of Flood Re, the supply of flood insurance in the UK was subject to a "Statement of Principles on the Provision of Flood Insurance" (SoP) agreed in 2002 between the UK government and the insurance industry, represented by the Association of British Insurers (ABI).<sup>7</sup> Under the SoP, insurers agreed to offer flood insurance to all residential properties without any restrictions on insurance premiums.<sup>8</sup> However, the development of more accurate flood risk models during the 2000s, the publication of public flood maps by the Environment Agency in 2004, and the increasing frequency of extreme floods raised concerns about the affordability of flood insurance (Belanger and Bourdeau-Brien, 2018). In 2013, the insurance industry and the government agreed to create a non-profit reinsurance scheme, Flood Re, to replace the SoP in 2016. As an interim solution, they extended the SoP for another three years.

In Section 11.1 of the Online Appendix, we compare Flood Re with the NFIP, which has been the subject of several studies (e.g. Georgic and Klaiber (2022); Ge et al. (2022), see also Introduction). Here we summarise the main differences. First, the NFIP provides subsidized flood insurance to households, while Flood Re provides reinsurance. Second, the NFIP provides insurance only in designated at-risk areas, where flood insurance is mandatory for mortgage borrowers, while UK insurers are free to pick which policies to cede to Flood Re. Third, the NFIP is funded by central government borrowing, while the main funding source for Flood Re is a levee on private insurers. Finally, Flood Re is designed to phase out.

#### 3 Data and Sample

#### 3.1 Data

To understand the effect of Flood Re on property prices, we employ four data sets. The first data set includes property transactions, the second and third contain the measures of flood risk, and the fourth data set includes the characteristics of local authority districts.<sup>9</sup>

 $<sup>^7\</sup>mathrm{In}$  turn, the SoP replaced previous, similar, informal arrangements between government and industry.

<sup>&</sup>lt;sup>8</sup>The agreement did not cover properties in the highest flood risk categories and properties built after 2009. <sup>9</sup>Local authority district is a level of administrative division of England. There are a total of 343 local authority districts in England, comprising five types of local authority: county councils, district councils, unitary authorities, metropolitan districts and London boroughs.

#### 3.1.1 HM Land Registry Price Paid Data

Price Paid Data (PPD) from HM Land Registry covers the universe of transactions of residential properties in England since 1995, and it has been used in several studies on the UK property market, e.g., Giglio et al. (2015); Bracke and Tenreyro (2021)). For each property, the dataset contains information on the exact address, the transaction date and the price, and characteristics such as property type (detached, semi-detached, flat), whether the property is a new-built, and ownership type (leasehold versus freehold).

#### 3.1.2 Recorded Flood Outlines

Our first measurement of flood risk is based on the Recorded Flood Outlines produced by the Environment Agency, which document historic flooding since 1946 as GIS layers. To match the layers with the property transaction data set, we map them to 6-digit postcode units, which cover a small area (15 properties on average). The data records the exact dates of the start and end of each flood outline, allowing us to calculate the duration of each flood event and the time interval between each property transaction and the latest flood event to affect the property.

We classify a property as "flooded" if at least one flood event lasting for more than a day is recorded in the four years before the transaction.<sup>10</sup> To isolate minor flood events, we identify a property as "flash-flooded" if only flood events lasting for up to a day are recorded in the four years before the transaction. The geographic distribution of "flooded" properties are depicted in Panel A of Figure 1.

#### 3.1.3 Flood Map

Our second measure of flood risk is extracted from flood maps published by the Environment Agency. Compared to actual flood events, our *ex-post* measure, flood maps provide an estimate of the *ex-ante* flood risk of properties. This measure is then aggregated at 6-digit postcode level, and provides information on the number of properties classified in four

 $<sup>^{10}\</sup>mathrm{We}$  motivate the four-year cut-off in Section 4.

flood-risk categories.<sup>11</sup> To simplify the analysis, we calculate the average flood risk for the 6-digit postcode and then classify all properties in 6-digit postcodes with an average annual flood probability of more (less) than 0.1% as a "at-risk" ("no-risk") property. The locations of the at-risk properties are shown in Panel B of Figure 1, showing that at-risk properties are clustered in areas that have been exposed to flood events (in Panel A).

The ex-ante flood risk measure is less related to physical damage from actual floods, while the ex-post measure could, at least in part, also reflect differences in the physical state of the building. The limitation of the ex-ante risk data is that they barely change over time. Although the Flood Map is updated annually, the variations across year are rather limited, apart from a major update in 2013-2014 (Garbarino and Guin, 2021). For our analysis, we use the 2016 version of this flood map. Comparing the ex-ante risk measure in 2016 with the latest measure in 2023, only 0.98% of the no-risk properties in 2016 are now classified as at-risk properties.

#### 3.1.4 Local authority characteristics

To examine the heterogeneous effects of Flood Re, we employ the English Indices of Deprivation which provide: an overall measure of the deprivation level; the proportion of population with National Qualifications Framework (NQF) level 4 or above qualification, e.g., degree with honours and postgraduate certificate; average income; age; percentage of rental properties of local authorities; rural-urban classification. We also use general election results recorded by the House of Common to measure the percentage of votes for the Green Party in local authorities in the 2019 United Kingdom general election, and EU referendum results as recorded by Data.gov.uk.

#### **3.2** Sample construction

To construct our sample, we start with the universe of all property transactions in England between 1995 and 2020. The first step of sample filtering addresses concerns over the change in

 $<sup>^{11}</sup>$  The categories are: very low (one-year ahead flood probability less than 0.1%), low (between 0.1% and 1%), medium (between 1% and 3.3%) and high (greater than 3.3%).

the planning process for new buildings after the publication of the Planning Policy Guidance Note 25 (PPG25) (DTLR, 2001). PPG25 required local planning authorities to employ a set of decision rules accounting for flood risk, and to consult with the Environment Agency on permissions to build in flood-risk areas. As a results, the rejection rate of building permissions on flood risk grounds increased from 10% in 2001 to 33% in 2004 (Porter and Demeritt, 2012). Properties built after the publication of the PPG25 are therefore likely to be more resilient to flood risk. To alleviate this concern, our sample excludes properties built after 2002.

In the second step, we select a subsample of properties that were transacted at least twice since 1995, with at least one transaction in the sample period which covers the four years before and after Flood Re. The relatively short sample period around the introduction of Flood Re mitigates the concern that our findings are driven by the improvement in flood defences over time.

We then convert the data into a panel structure by identifying the series of transactions of the same property, using the full address. After taking first differences of the transaction price, the sample comprises 1,754,067 observations of 1,563,062 properties. We then match this repeat-transactions sample with the two flood risk measures using the 6-digit postcode, and with the local authority-level data using the local authority identifier.

#### 3.3 Summary statistics

Table 1 shows summary statistics for our sample.<sup>12</sup> In Panel A, we present summary statistics for property-level variables. The average property price in our sample is £226,840 with a growth rate of 42.4% between transactions. The appreciation of properties is rather large because of the long time interval between transactions, which is around eight years and four months. A small proportion of properties are newly built at the time of the first transaction in our sample. The majority of properties in the sample is detached, semi-detached or terraced, and around 15% of them are flats. The large majority of the properties are freehold and the remaining are leasehold. In Panel B, we show summary statistics for flood risk measures.

<sup>&</sup>lt;sup>12</sup>Definitions of variables are detailed in Table A.1 in Online Appendix.

Around 0.3% of the properties experience at least one severe flood event (lasting more than a day) in the four years before a transaction, and 0.1% experience flash floods (lasting less than for a day) in the four years before a transaction. About 11% of the properties are classified as at-risk properties in terms of the annual probability of being flooded. In Panel C, we summarize the eight local-authority level characteristics used for heterogeneity analysis.

#### 4 Empirical strategy

To identify the effect of flood risk on property prices and the mitigating effect of Flood Re, we estimate the following equation:<sup>13</sup>

$$\Delta ln(Price_{i,g,t}) = ln(Price_{i,g,t}) - ln(Price_{i,g,t'}) = \beta_0 + \beta_1 Flood Risk_{i,g,t} + \beta_2 Flood Risk_{i,g,t} \times Post Flood Re_t + \gamma X_{i,g,t} + \delta_{g,t} \times \delta_{g,t'} + \varepsilon_{i,g,t}$$
(1)

where  $\Delta ln(Price_{i,g,t})$  is the outcome variable, calculated as the difference between  $ln(Price_{i,g,t})$ , the natural logarithm of the value of the property *i* in 3 digit postcode *g* in year *t* of the latest transaction, and  $ln(Price_{i,g,t'})$ , the natural logarithm of the value of the same property in the previous transaction year t'.

Flood Risk<sub>i,g,t</sub> indicates flood risk of property *i* and the related coefficient  $\beta_1$  captures the effects of flood risk on property prices before the introduction of Flood Re. We employ different flood risk indicators. The primary measurement is a dummy variable, Flooded<sub>i,g,t</sub>, which indicates whether the property experiences at least one flood event lasting for more than a day in the four years before the transaction in year *t*. The dummy variable Flash Flooded<sub>i,g,t</sub> equals one if the property only experiences flood events lasting up to a day in the four years before the transaction in year *t*. For both Flooded<sub>i,g,t</sub> and Flash Flooded<sub>i,g,t</sub>, we use the 4-year cut-off to reflect on Beltrán et al. (2019) who find that prices of properties affected by flooding decline for merely 4 years. In Figure A.2 in our Online Appendix, our pre-Flood Re observations show a similar pattern, confirming that the negative effect of flooding on property prices vanishes after 4 years. Flood events are an ex-post measure

<sup>&</sup>lt;sup>13</sup>In the Online Appendix, we provide a simple conceptual framework which motivates the empirical strategy discussed in this section.

of flood risk and could, at least in part, also reflect differences in the physical state of the building due to flood damage. The high penetration rate of flood insurance in the UK (see Section 2) limits concerns that owners of damaged properties might sell before the damage from flooding has been rectified. An alternative, ex-ante risk measure that is less likely to reflect flood damages come from the Environment Agency's flood risk maps. We use the 2016 version to construct a dummy variable, At-risk<sub>i,g</sub>. It indicates if the property has, according to the Environment Agency risk classification, an annual flood risk probability larger than 0.1%. Whilst this measure mitigates concerns about flood damage, it comes with the limitation of barely changing over time.<sup>14</sup> Overall, the two measures are correlated, since the ex-ante risk measure reflects the probability of flooding.<sup>15</sup> We therefore present the ex-post and ex-ante variables as complementary measures in the main results of our paper.

Post Flood  $Re_t$  is a dummy variable indicating whether the property transaction in year t happened after the implementation of Flood Re. The interaction term,  $Flood Risk_{i,g,t} \times Post Flood Re_t$ , is our variable of interest and the related coefficient,  $\beta_2$ , captures the effect of Flood Re on the prices of flooded/at-risk properties. The impact of flood risk on property prices after Flood Re is therefore measured by the sum of  $\beta_1$  and  $\beta_2$  in equation 1. A negative  $\beta_1$  and a positive  $\beta_2$  with similar magnitude support the conjecture that Flood Re mitigates the negative effect of flood risk on property prices.  $X_{i,g,t}$  is a vector of control variables, reflecting property characteristics, i.e. property type, year of construction and form of tenure (freehold vs. leasehold).

 $\delta_{g,t}$  and  $\delta_{g,t'}$  are fixed effects for each combination of 3-digit postcode and transaction year, where t is the year of the last transaction, and t' the year of the previous transaction.  $\delta_{g,t}$  and  $\delta_{g,t'}$  capture further confounding factors, such as the supply of new properties, affecting property values in the 3-digit postcode areas in the years of latest and previous transaction.<sup>16</sup> The interaction  $\delta_{g,t} \times \delta_{g,t'}$  allows us to isolate the effects of flood risk and Flood

<sup>&</sup>lt;sup>14</sup>Comparing the ex-ante risk measure in 2016 with the latest measure in 2023, only 0.98% of the no-risk properties in 2016 are now classified as at-risk properties.

 $<sup>^{15}</sup>$  Within our sample 1.9% of the properties classified as ex-ante high risk were flooded, compared to 0.2% of properties classified as low ex-ante risk.

<sup>&</sup>lt;sup>16</sup>Each 3-digit postcode contains on average around 6,000 properties.

Re on flood-prone properties from other confounding factors and price trends that might influence the value of properties in the same postcode and transacted in the same years. In an alternative specification, we use additive (instead of interacted) fixed effects to relax the condition that the transactions must happen in exactly the same two years. As a result, the fixed effects are less demanding in the time dimension whilst we keep the geographical dimension constant. Finally,  $\varepsilon_{i,g,t}$  is the error term. We cluster standard errors at the local authority level.

Equation 1, with ex-post and ex-ante risk measures, is also the basis for robustness checks and heterogeneity analyses. Most of the analysis focuses on property prices, but in Section 7.1 we look at the effect of Flood Re on transaction probabilities. The specification for transaction probabilities is similar to the one used for property prices but the outcome variable is a dummy indicating whether a property was transacted in a specific year.

Our empirical strategy follows a Difference-in-Difference (DiD) framework. Properties exposed to flood risk - based on either ex-post or ex-ante measures - form the treatment group whilst those that are not exposed form the control group. In our empirical analyses, we examine their change in housing prices relative to the timing of Flood Re. Identification relies on the assumption that properties not exposed to flood risk serve as an appropriate control group, comparable to those that are at risk. To assess the extent to which differences between treatment and control groups could be a concern for identification, we compare observable property characteristics based on their exposure to flood risk. Panel A of Table 2 presents mean values according to ex-post flood exposure. Similarly, Panel B compares properties in at-risk areas to those in no-risk areas using our ex-ante measure of flood risk. In both panels, we compute normalized differences between groups across various property-level and local authority-level characteristics. The normalized differences for all characteristics are below the critical threshold of 0.25, as suggested by Imbens and Wooldridge (2009) and Imbens and Rubin (2015). The only exception is the rental rate, which is lower in flooded areas compared to non-flooded areas. Overall, the results indicate that our findings are unlikely to be driven by heterogeneous property and demographic characteristics across areas with varying levels of flood risk.<sup>17</sup> We also address potential self-selection of properties in our house price sample. In Section 7.1 we show that flooded properties are more likely to be transacted after Flood Re, which is consistent with Flood Re reducing buyers' concerns about flood insurance premiums.<sup>18</sup>

A further identification concern is that the introduction of Flood Re might have been anticipated by home buyers, and reflected in transaction prices, already before its official start in 2016. To that end, we examine the effect of the extension of the Statement of Principles (SoP) in 2013. This extension coincided with the announcement that the SoP would be replaced by Flood Re in 2016. Our findings, presented in Section 5.1, do not suggest announcement effects on property prices. To assure us of the validity of the DiD parallel trends assumption, we also provide estimates of the annual treatment effects over time. House price trends appear parallel before the introduction of Flood Re, which we discuss in Section 11.3 of the Online Appendix.

#### 5 Effect of flood events and Flood Re on property prices

Panel A of Table 3 summarises the estimation results of the main specification in equation (1). Columns 1-3 show results using the ex-post measure of flood risk based on actual flood events (*Flooded*), and Columns 4-6 present results using the ex-ante measures based on Environment Agency risk flood maps (At-risk).

We first provide a benchmark for the effect of flood risk on property prices: In Column 1, we present results of a simplified version of equation 1 using only the ex-post measure of flood risk,  $Flooded_{i,g,t}$ , but not yet interacting it with our Flood Re indicator,  $Post Flood Re_t$ .

<sup>&</sup>lt;sup>17</sup>Concerns about selection of households or properties into high flood risk areas are less relevant in our regression analyses. Whilst Table 2 presents property and local authority characteristics across different local authorities, our regression framework does not exploit cross-local authority level variation but it leverages even more granular, within 3-digit postcode variation by employing geographic fixed effects. Therefore, in our regressions using ex-post flood events, we effectively control for the ex-ante risk that is constant in a small geographic scope. The flood events variable captures mainly the unanticipated effect of flooding.

<sup>&</sup>lt;sup>18</sup>Selection of properties into treatment and control groups is not possible. The properties in our sample were built several years before the introduction of Flood Re, and the risk measures can only change with local area, not individual property, characteristics.

We use simple, additive fixed effects but we do not employ any property level control variables. The results confirm previous findings (Lamond and Proverbs, 2006; Kousky, 2010; Bernstein et al., 2019), indicating that flooded properties experience a 0.9% (*t*-statistics -2.21) decrease in property prices, while there is no effect for less severe flash floods. Similarly, Column 4 presents the estimation results with the ex-ante flood risk measure, At-risk<sub>i,g</sub>. At-risk properties are sold at a discount of a 0.3% (*t*-statistics -3.49). The effect is lower than actual flood events, a result that is consistent with recent findings in Niu et al. (2023) showing that ex-ante flood risk is less reflected on property prices due to its low salience.

We then introduce the variable  $Post Flood Re_t$  into the estimations to differentiate the effect of flood risk after the introduction of Flood Re. The interaction term,  $Flooded_{i,g,t} \times Post Flood Re_t$  (At-risk<sub>i,g</sub> × Post Flood Re<sub>t</sub>), indicates whether Flood Re plays a role in mitigating the negative effect of flood risk, with a positive coefficient suggesting that it mitigates the effect of flood risk on property values. We also expect the introduction of the interaction term  $Flooded_{i,g,t} \times Post Flood Re_t$  to change the magnitude of the coefficient of  $Flooded_{i,g,t}$ , increasing it compared with the results in Column 1 and 4, as it now reflects the effect of flood risk before the introduction of Flood Re. Estimation results are presented in Columns 2 (using flood events) and 5 (using our ex-ante flood risk measure). In these specifications, we also add property controls, but the fixed effects are still additive. In Column 2, flood events longer than a day reduce property values by 1.5% (t-statistics -2.70) before Flood Re, mitigated by 1.4% with the introduction of Flood Re. The net negative effect of flood events on prices (calculated by adding these two coefficients) is only 0.1% after the introduction of Flood Re. Consistent with the findings in Column 1, there is no evidence that flash floods affect property prices in either periods (before and after the implementation of Flood Re). In Column 5, we find that at-risk properties are sold at a 0.4% (t-statistics -3.24) discount before the introduction of Flood Re but the discount disappears after the introduction of Flood Re. In short, the introduction of Flood Re completely offsets the negative pricing effect of flood risk, irrespective of the risk measure we employ.

Columns 3 and 6 show the estimation results with our preferred specification in equation 1, with interacted postcode × year fixed effects for each transaction ( $\delta_{g,t}$  and  $\delta_{g,t'}$ ). The results are similar to those presented Columns 2 and 5. Column 3 (6) shows that flooded (at-risk) properties experience a 1.6% (0.4%) drop in value. The estimated coefficient of the variable, *Flooded*<sub>*i,g,t*</sub> × *Post Flood* (*At-risk*<sub>*i,g*</sub> × *Post Flood*), is 1.8% (0.4%), suggesting that flood events (ex-ante flood risk) do not reduce property values after the implementation of Flood Re. In short, the results with interacted fixed effects are similar to those with additive fixed effects (Columns 2 and 5), addressing concerns that interacted fixed effects might be too demanding and thereby over-controlling. In the remainder of the paper, we will present the specification using interacted fixed effects.<sup>19</sup>

In Panel B of Table 3 we replace the binary risk variables with continuous measurements for flood risk, namely flood duration (for the ex-post measure) and the annual probability of flooding (for the ex-ante measure). We conjecture that the negative effects on property prices increases with the level of flood risk, and, as a result, properties that are more prone to flooding are expected to benefit more from Flood Re. The specifications in Panel B mirror the respective columns in Panel A. The negative effects on property prices indeed increase with the duration of flood events and the ex-ante flood risk. Importantly, the mitigating effects of Flood Re on property values also increase with the level of flood risk. Results presented in column 3 of Panel B suggest that before the introduction of Flood Re, a 100-days increase in the duration of the flood reduces property prices by an additional 2.4%.<sup>20</sup> For ex-ante risk, we use the mid-point flood probability in each risk bucket. Before the introduction of Flood Re, a 1 percentage point increase in ex-ante flood risk reduced property prices by an additional 0.1%.<sup>21</sup> In both cases, the negative effects of flood risk are offset after the introduction of Flood Re.

<sup>19</sup>In Section 11.3 and Figure A.3 of the Online Appendix, we shed light on the dynamic impact of Flood Re by employing the event-study approach.

 $<sup>^{20}</sup>$ The mean duration of a flood in our sample is 20 days.

<sup>&</sup>lt;sup>21</sup>The mean ex-ante risk of at-risk properties in our sample is 1.4%.

#### 5.1 Falsification tests

To examine whether property prices are indeed affected by Flood Re, we conduct two sets of falsification tests. The first test relates to the introduction of Flood Re and implicitly also tests whether there are announcement effects. We redefine the sample to property transactions in April 2010 to April 2016 and use the extension of the Statement of Principle (SoP), briefly described in Section 2, in July 2013 as a placebo treatment to flooded properties. As the extension did not change the terms of SoP, it should not affect the price of flooded properties. We replace the variable *Post Flood Re<sub>t</sub>* in equation 1 with *Post SoP extension<sub>t</sub>*, which equals to 1 if the transaction is after July 2013 (0 otherwise). This specification estimates how the SoP extension affects flooded property prices. Columns 1 and 2 in Panel A of Table 4 show that the interaction term is not different from zero, suggesting that value of flooded properties is unaffected by the placebo treatment. Since the extension of the SoP coincided with the announcement that Flood Re would become fully operational in 2016, these results also suggest that there was no effect of the announcement on property prices.

In the second falsification test, we verify the effect of the flood treatment. We constrain the sample to properties that are not being flooded in the past four years of transactions. We employ the genuine Flood Re introduction date but we randomly assign properties to be "flooded" properties and replicate the estimation equation 1. We then run Monte Carlo simulations with 1,000 replications of equation 1 to check whether non-flooded properties are affected by Flood Re. As Flood Re should not affect properties that are not at flood risk, the null of zero effect is true. Thus, we should only reject the null by making Type 1 errors. Panel B of Table 4 shows that the rejection rates are in line with those that would occur through Type 1 errors. In most cases, the average value the coefficients of *Pseudo flood*<sub>i,g,t</sub> and *Pseudo flood*<sub>i,g,t</sub> × *Post Flood Re*<sub>t</sub> are close to 0, suggesting that non flooded properties are unaffected by Flood Re.

#### 5.2 Robustness tests

We perform four robustness tests, shown in Table 5. First, the estimated coefficients could be biased if flood events change the hedonic price function of property. For example, flood events may induce buyers to buy newer properties with better protection against flooding. To account for such possibility, we follow Kuminoff and Pope (2014) to control for the interactions of flood event indicators and various property characteristics. The results in Column 1 show that the inference of our estimated coefficients of interest is the same as the baseline results.

Second, to address the correlation between ex-ante flood risk and flood events, we replicate our estimation with both the ex-ante flood risk and flood events as explanatory variables. The results (shown in Column 2) are consistent with the literature which suggests that flood events are more salient and has stronger negative impact on house price than long run flood risk, e.g., Atreya et al. (2013); Zhang and Leonard (2019). The results highlight that recently-flooded properties benefit more from the mitigating effect of Flood Re.

Third, in Column 3 we implement a robustness test that includes only postcodes with at least one flooded property. The estimated coefficients are similar to those in our main results in Table 3. Finally, in our sample, about 16% of the properties are flats. Flood events might have different impact on flats, because some flats are above ground floor. To alleviate such concerns, we replicate the estimation in a sample excluding flats. The results are shown in Column 4. The inference of our estimated coefficients of interest stays the same, suggesting that the estimation of our key coefficients is not biased by the inclusion of flats.

#### 6 Heterogeneous effects of Flood Re

Flood Re might exert heterogeneous effects on different sub-populations as they might react differently to the risk of flooding. In this section, we examine whether Flood Re has different effects in subsamples that reflect heterogeneity in property values and in the socioeconomic characteristics of local areas. First, we replicate the estimation in Column 3 of Table 3 with subsamples constructed using the distribution of property prices in the first transaction. Figure 2 shows the estimated coefficient and 95% confidence interval of the variable *Flooded*<sub>*i*,*g*,*t*</sub> x *Post Flood Re*<sub>*t*</sub> for each quintile of the distribution of property prices. Flood Re has a stronger effect on more expensive properties (properties whose value is higher than the  $60^{\text{th}}$  percentile (*p*60) of property prices in the sample) and has a limited effect on lower-value properties (properties whose value is lower than or equals to the *p*60 of property prices).

To provide a richer picture of the heterogeneous effects of Flood Re, we complement our property transaction data with a number of local-authority level indicators on income, deprivation, age, education, urban versus rural location, and the share of rental properties. We split the sample based on the median value of each indicator (apart from the urban/rural indicator) and replicate the estimation of equation 1. Table 6 shows results using the ex-post measure of flood risk.<sup>22</sup> Columns 1 and 2 present estimation results for properties in local authorities with average income above and below the median, respectively. The results suggest that a) the negative effect of flood events on property prices is stronger in local authorities that have higher average income, b) households in higher income areas benefit more from Flood Re. The Chow test *F*-statistics verify that the coefficients of the two groups are significantly different at 5% significance level.

Income alone is, however, an incomplete measure of local living standards (Ringen, 1987, 1988). Columns 3 and 4 show results with sample splits using an index of multiple deprivation that, in addition to income, takes into account six other domains, including employment, education, health, crime, barriers to housing and local services, and living environment.<sup>23</sup> The negative effect of flood events on property prices is stronger in *less* deprived local authorities, and these areas benefit more from Flood Re. The Chow test *F*-statistics indicates that the coefficients of the two groups are significantly different at 5% significance level.

We also look at heterogeneity according to demographic characteristics of the local authority, including average age, average education level, urban/rural location and the share of rented properties. Detailed results are shown in Columns 5 to 12 in Table 6. Here, we

<sup>&</sup>lt;sup>22</sup>Results with the ex-ante measure are similar and are available in the Online Appendix in Table A.5. The correlation matrix between the socioeconomic indicators is shown in Table A.2.

<sup>&</sup>lt;sup>23</sup>See Payne and Abel (2012) for more details of the background and computation method of the English indices of deprivation.

summarise them briefly. The effect of flood events is similar across areas with above versus below average age, but the effect of Flood Re is stronger in areas with older populations (Columns 5 and 6). The Chow test *F*-statistics verifies that the coefficients of the two groups are significantly different at 5% significance level. Areas with a more educated population have a stronger effect of Flood Re on flooded properties value (Columns 7 and 8). However, the Chow test *F*-statistics suggests that the difference is statistically insignificant at 10% level. The effects of flood events and of Flood Re are stronger in urban areas (Columns 9 and 10). The Chow test *F*-statistics suggests that the coefficients of the two groups are significantly different at 1% significance level. Finally, the effect of flood is similar in areas with high versus low share of rental properties (Columns 11 and 12), but Flood Re only mitigates the negative effect on property price in areas with a lower share of rental properties. The Chow test *F*-statistics suggests that the coefficients of the two groups are significantly different at 1% significance level.

Together, these results suggest that Flood Re has distributional implications. It appears to disproportionately benefit (in terms of the appreciation of flood-prone properties' values) owners of more expensive properties living in areas with higher income and lower deprivation. One possible interpretation for this result is that socioeconomic differences capture differences in financial sophistication and awareness of climate risks (Fielding and Burningham (2005); Fielding (2012)).<sup>24</sup> Older households with higher education levels and living in urban areas might be more aware of the implications of flood risk and Flood Re for property prices. Some support for this interpretation is provided by survey data indicating that older households living in flood-prone areas are more likely to be aware of Flood Re.<sup>25</sup> The heterogeneity results also appear consistent with previous findings sowing that the value of rural properties

<sup>&</sup>lt;sup>24</sup>While the difference in renovation rate could probably explain the heterogeneous effects of Flood Re in different social classes, Garbarino and Guin (2021) show that flood events in the UK do not affect the renovation rate of residential properties. Therefore, renovation after flooding may not be a major factors driving our results. Also, the UK has number of home renovation subsidisation programmes for low-income households, such as the Green Homes Grant Support and Support for Mortgage Interest (SMI) for home improvement. The availability of these programmes mitigate the possibilities that the properties of low-income households being severely under-renovated.

<sup>&</sup>lt;sup>25</sup>We employ the survey data from the 2018 Availability and Affordability of Insurance report conducted by DEFRA to examine the correlation between different demographic characteristics and the awareness of Flood Re (Defra, 2018). The results are presented in Table A.3 in the Online Appendix.

is less affected by flood events (Beltrán et al., 2019), and that renters are less sensitive to future inundation risk than homeowners (Bernstein et al., 2019).

Previous studies have also shown that heterogeneous beliefs about risks from climate change affect property values. Baldauf et al. (2020) find that the value of properties exposed to climate risks are more likely to sell at discount if they are located in areas with more climate change "believers". However, beliefs in climate risks may play a limited role in our analysis if insurance premium discounts from Flood Re are salient to households. To examine these two opposing hypotheses, we employ the percentage of votes for the Green Party in the 2019 United Kingdom general election as a proxy for the differences in belief of climate change risk across local authorities. If awareness of climate risks is the driver of the heterogeneous effects, the effect of flood and Flood Re is expected to be stronger in local authorities with higher share of votes to the Green Party. Columns 1-2 in Table 7 present the estimation results. The Chow test suggests that there is no significant difference across local authorities with Green Party votes above versus below the median, and the coefficients of the two key variables, *Flooded*<sub>i,g,t</sub> and *Flooded*<sub>i,g,t</sub> × *Post Flood Re*<sub>t</sub>, are similar across the two groups.<sup>26</sup>

We use the voting results in the 2016 "Brexit" referendum on UK membership in the European Union as an alternative measurement of local views on climate change. Survey findings indicate that "Leave" voters are much less likely to believe in climate change risks than "Remain" voters.<sup>27</sup> The results in Columns 3 and 4 in Table 7 show a stronger impact of Flood Re in areas with a higher share of "Leave" vote, yet the differences in coefficients among the two sub-group are statistically insignificant at 10% significance level. Overall, these results imply that differences in concerns about climate risks do not help explaining the heterogeneous effects of Flood Re.

 $<sup>^{26}</sup>$ Results using the ex-ante measure are available in Table A.6 of the Online Appendix.

<sup>&</sup>lt;sup>27</sup>Details of the survey, conducted by Savanta ComRes, can be found on https://comresglobal.com/polls/assaad-razzouk-eu-referendum-and- science-poll/.

#### 7 Mechanisms

In this section, we enhance the interpretation of the effect of Flood Re on the property market by assessing three potential mechanisms. First, we use transaction data to disentangle demand vs supply-side interpretations of price changes. Second, we interact ex-ante and ex-post measures of flood risk to evaluate surprise vs. confirmation effects. Third, we consider properties built after 2002 to provide evidence on the role of tightening building standards in flood-prone areas.

#### 7.1 Demand vs supply of properties

The price discount for flood-prone properties could be related to buyers' reluctance to purchase at-risk properties (a reduction in demand) or current owners' desire to sell and move out of risky areas (increasing the supply of properties on the market). In the former case, we should observe a decrease in property transactions in flooded areas. In the latter, we expect to see an increase. In either case the effect of flood risk on transactions should be mitigated by the introduction of Flood Re.

To verify if our findings are driven by demand or by supply, we examine the changes in transaction volumes accompanying flood events and the introduction of Flood Re. Following Bernstein et al. (2019), we expand the original sample into a balanced panel data set in which each property has an observation in each year of the sample period.<sup>28</sup> This allows us to estimate the following equation:

$$Trade_{i,g,t} = \beta_0 + \beta_1 Flooded_{i,g,t} + \beta_2 Flooded_{i,g,t} \times Post Flood Re_t + \beta_3 Flash Flooded_{i,g,t} + \beta_4 Flash Flooded_{i,g,t} \times Post Flood Re_t + \gamma X_{i,g,t} + \delta_{g,t} + \varepsilon_{i,g,t}$$

$$(2)$$

where  $Trade_{i,g,t}$  is the outcome variable, indicating whether the property is traded in year t,  $Trade_{i,g,t}$  is one if property i is traded in year t, 0 otherwise.  $\delta_{g,t}$  captures the confounding factors affecting the property of being traded in the 3-digit postcode g in year t. Definitions

<sup>&</sup>lt;sup>28</sup>The sample still includes the same properties used to model price effects, i.e. properties that were transacted at least twice since 1995 and at least one transaction is in the sample period which covers the four years before and after Flood Re.

of other variables follow equation 1.

In Column 1 of Table 8, we show the estimation results for equation 2. Flooded properties are 3.6% less likely to be transacted in the following four years (from a base transaction rate of 15.3%). Flood Re not only mitigates the negative effect but it increases the transaction probability by 2.4%. The results plausibly reflect the sales of the accumulated properties that were flooded before the introduction of Flood Re. We also find that flooded properties are being traded later than non-flooded properties, and Flood Re completely mitigates this effect. The results are shown in Column 2 of Table 8. We interpret this result with caution, as it plausibly suffers from reverse causality, i.e. longer time interval between transactions increase the likelihood of being flooded. Overall, lower transaction probabilities in the period before Flood Re suggest that the price discount for flood-prone properties is driven by a reduction in demand. A more in-depth analysis of demand and supply behaviour would require data on property listings (Genesove and Mayer, 2001) or on price expectations (Armona et al., 2019; Bakkensen and Barrage, 2022).

The results in Table 8 also help address potential concerns with regards to selection in the sample of transacted properties. In theory, the price effect could be explained by differences in unobserved risk characteristics between the pre- and post- Flood Re periods. If properties transacted after Flood Re are lower risk, then we should expect higher relative prices. But we find that flooded properties are more likely to be transacted after Flood Re, suggesting that the selection effect goes in the opposite direction, with riskier properties more likely to be transacted after Flood Re. This result is consistent with Flood Re reducing buyers' concerns about flood insurance premiums.

#### 7.2 Surprise versus confirmation effects

Flood events can have different effects on house prices depending on properties' risk exposure. The negative price effect of floods events could be stronger in no-risk areas (as floods are more surprising) or in at-risk areas (as new floods reinforce existing information about flood risk). We examine the interaction between ex-ante and ex-post indicators for flood risk in Table 9. In Columns 1 and 2, we run the specification with ex-post flood indicators but split the sample between at-risk and no-risk areas. The effect is concentrated in at-risk areas, where flooding reduces prices by 2.5% before the introduction of Flood Re, and we see a positive coefficient (3.4%) on the interaction term for the post Flood Re period. In no-risk areas, the negative effect is small and statistically insignificant in the pre-Flood Re period, and so is the coefficient on the interaction term for the post Flood Re period. Columns 3 and 4 show similar results for trade probabilities. The negative effect of flooding on trading is larger in high ex-ante risk relative to low ex-ante risk areas (-5.2% vs. -3.1%), both in both cases the effect is statistically significant. The positive offsetting effect of Flood Re for flooded properties is larger in at-risk relative to no-risk areas (8.9% vs. 5.2%).

Flooding in at-risk areas seems to lead to a larger reassessment of property prices than in no-risk areas. In other words, the 'confirmation' effect of flooding in at-risk areas appears stronger than the 'surprise' effect in no-risk areas. This could be because flood risk becomes more salient, or because of an actual increase in premiums, or a combination of the two.

#### 7.3 Building standards

As described in Section 3.2, so far we have excluded from the sample properties built after 2002, when tighter building standards were introduced for new developments in flood-prone areas. We expect these properties to be more resilient to flooding, and to benefit less from the introduction of Flood Re. To test this, we reintroduce properties built after 2002 in our sample. There are about 250,000 such properties, compared to about 1,750,000 properties built pre-2002. In Table 10, we compare results for properties built up to 2002 (Column 1), our baseline results, against results for properties built after 2002 (Column 2). For properties built after 2002, we observe neither a decrease in prices as a result of flooding prior to the introduction of Flood Re nor an offsetting positive effect afterwards.<sup>29</sup> To limit incentives to build new properties in flood-prone areas, Flood Re does not cover properties built after 2009. In Column 3 of Table 10 we show results for the post-2009 subset of properties. The

<sup>&</sup>lt;sup>29</sup>In Table A.7 of the Online Appendix, we show that employing ex-ante risk measure do not affect the inference of the results.

coefficients for flooded properties and the interaction term are similar to those for the whole subset of post-2002 properties, and are also not statistically significant.

Overall, these results are consistent with the idea that properties built after 2002 are more resilient to flood risk, and that this is reflected in their pricing. The exclusion of properties built after 2009 from the Flood Re cover does not affect their pricing. From a policy perspective, these results also suggest that tighter building standard might limit the need for reinsurance schemes for new properties.

#### 8 Housing wealth effects

It is not the aim of this paper to offer a comprehensive assessment of the costs and benefits of the Flood Re. We can however provide a back-of-the-envelope comparison of the scheme's costs with its overall effects on housing wealth. First, we assume that the average property price is £226,840 (as in our sample) and that there are about 5.2 millions at-risk properties in England (as estimated by the Environment Agency). In our baseline calculation, we use the coefficient of the interaction term in our preferred specification in Panel A of Table 3, which indicates a 1.8% increase in the price of flooded properties after the introduciton of Flood Re. In monetary terms, the average price effect is £226,840 × 1.8% = £4,083. Assuming that all 5.2m at-risk properties are on 100-year flood plain (i.e. 1% annual flood probability), this translates to a net wealth effect of 5.2m properties at risk × 1% risk × £4,083 = £212.3m.<sup>30</sup> This estimate of net wealth effect of Flood Re is in a similar order of magnitude but above the annual costs of the Flood Re scheme, which is a levy of £180 for the UK, broken down to about £151m in England.<sup>31</sup>

<sup>&</sup>lt;sup>30</sup>The Environment Agency does not specify the average annual flood probability for properties in flood plains. A more conservative assumption on flood risk would result in higher net wealth effects. E.g., taking the average property price of at-risk properties, which is £293,514, or employing the average risk in the data, which 1.4% based on our mid-point measure, the net wealth effect increases.

<sup>&</sup>lt;sup>31</sup>Assuming that about 84% of all dwellings are located in England, the pro-rata levy for England would be £151m. These estimations are based on population in 2011, which is included in the sample period. Source: https://www.ons.gov.uk/peoplepopulationandcommunity/populationandmigration/ populationestimates/bulletins/annualmidyearpopulationestimates/mid2022

We can also employ our estimates to compare the net housing wealth effect in high-income versus low-income areas. Therefore, we calculate the average per-household wealth benefit of Flood Re relative to the average income per area (in %). To that end, we use the estimates of sub-sample analyses shown in Columns 1 & 2 of Table 6, which indicate a price effect of 1.9% (0.07%) for high-income (low-income) areas. Similarly, we know that the average property price is £364,712 (£195,943) and average income is £48,812 (£36,248) respectively. These numbers suggest a relative housing wealth effect in high-income areas of £364,712 × 1.9% ÷ £48,812 = 14.20% of the average income. Similarly, we obtain a relative housing wealth effect in low-income areas of 3.78% of average income. Even taking into account average costs per household relative to income, around 0.02% (0.03%) in high-income (low-income) areas, the net benefits of Flood Re appear to be concentrated in high-income areas.<sup>32</sup>

Beyond the housing wealth effects, there are other potential benefits from Flood Re that we cannot take into account. Flood Re results in lower insurance premiums, which may mitigate adverse selection among at-risk homes, addressing market frictions and improving insurance accessibility (Einav and Finkelstein, 2023). This can lead to enhanced post-disaster recovery and reduced strain on public finances (Unterberger, 2018). Moreover, insurance is often expensive for correlated tail risks like flooding, which can justify a centralized reinsurer with lower funding costs (Kousky and Cooke, 2012). These positive externalities may be not be reflected in house prices.

#### 9 Conclusion

In this paper, we examine how the introduction of a public, subsidized, reinsurance scheme affects the values of properties exposed to flood risk. Our results suggest that public reinsurance mitigates the negative effect of flood risk on property prices. Surprisingly, Flood Re has a weak impact in lower income and more deprived areas. These results have relevance for the debate on public flood insurance schemes beyond our specific UK setting, and are related, for example, to research on the capitalisation of insurance subsidies under US National

 $<sup>^{32}</sup>$ To calculate the average costs per household, we divide the levy per policy, which is estimated to be £10.5 (Surminski, 2018), by average household income in high-income (low-income) areas.

Flood Insurance Program (Nyce et al., 2015; Georgic and Klaiber, 2022; Ge et al., 2022). Our results suggest that such subsidies may matter for higher income areas and households, and the reason may be that they are more aware of the implications of climate risks for property prices.

#### References

- Armona, L., Fuster, A., and Zafar, B. (2019). Home price expectations and behaviour: Evidence from a randomized information experiment. *The Review of Economic Studies*, 86(4):1371–1410.
- Atreya, A., Ferreira, S., and Kriesel, W. (2013). Forgetting the flood? an analysis of the flood risk discount over time. *Land Economics*, 89(4):577–596.
- Bach, L., Calvet, L. E., and Sodini, P. (2020). Rich pickings? risk, return, and skill in household wealth. American Economic Review, 110(9):2703–47.
- Bakkensen, L. A. and Barrage, L. (2022). Going underwater? flood risk belief heterogeneity and coastal home price dynamics. *The Review of Financial Studies*, 35(8):3666–3709.
- Baldauf, M., Garlappi, L., and Yannelis, C. (2020). Does climate change affect real estate prices? only if you believe in it. *The Review of Financial Studies*, 33(3):1256–1295.
- Belanger, P. and Bourdeau-Brien, M. (2018). The impact of flood risk on the price of residential properties: the case of england. *Housing Studies*, 33(6):876–901.
- Beltrán, A., Maddison, D., and Elliott, R. (2019). The impact of flooding on property prices: A repeat-sales approach. *Journal of Environmental Economics and Management*, 95:62–86.
- Beltrán, A., Maddison, D., and Elliott, R. J. (2018). Is flood risk capitalised into property values? *Ecological Economics*, 146:668–685.
- Benjamin, J., Chinloy, P., and Jud, D. (2004). Why do households concentrate their wealth in housing? *Journal of Real Estate Research*, 26(4):329–344.
- Bento, A., Freedman, M., and Lang, C. (2015). Who benefits from environmental regulation? Evidence from the Clean Air Act Amendments. *Review of Economics and Statistics*, 97(3):610–622.
- Bernstein, A., Gustafson, M. T., and Lewis, R. (2019). Disaster on the horizon: The price effect of sea level rise. *Journal of Financial Economics*, 134(2):253–272.

- Bhatia, K. B. (1987). Real estate assets and consumer spending. The Quarterly Journal of Economics, 102(2):437–444.
- Billings, S. B., Gallagher, E. A., and Ricketts, L. (2022). Let the rich be flooded: the distribution of financial aid and distress after hurricane harvey. *Journal of Financial Economics*.
- Bin, O., Bishop, J., and Kousky, C. (2017). Does the national flood insurance program have redistributional effects? *The BE Journal of Economic Analysis & Policy*, 17(4).
- Bin, O., Bishop, J. A., and Kousky, C. (2012). Redistributional effects of the national flood insurance program. *Public Finance Review*, 40(3):360–380.
- Bin, O., Kruse, J. B., and Landry, C. E. (2008). Flood hazards, insurance rates, and amenities: Evidence from the coastal housing market. *Journal of Risk and Insurance*, 75(1):63–82.
- Bracke, P. and Tenreyro, S. (2021). History dependence in the housing market. American Economic Journal: Macroeconomics, 13(2):420–43.
- Campbell, J. Y. (2013). Mortgage market design. *Review of finance*, 17(1):1–33.
- Chaney, T., Sraer, D., and Thesmar, D. (2012). The collateral channel: How real estate shocks affect corporate investment. *American Economic Review*, 102(6):2381–2409.
- Crick, F., Jenkins, K., and Surminski, S. (2018). Strengthening insurance partnerships in the face of climate change–insights from an agent-based model of flood insurance in the uk. *Science of the total environment*, 636:192–204.
- da Silva Freitas, L. F., de Santana Ribeiro, L. C., de Souza, K. B., and Hewings, G. J. D. (2016). The distributional effects of emissions taxation in brazil and their implications for climate policy. *Energy Economics*, 59:37–44.
- DEFRA (2013). Flood insurance pricing, affordability and availability.
- Defra (2018). Availability and affordability of insurance July 2018. (July).
- DTLR (2001). Planning policy guidance note 25: Development and flood risk.

- Einav, L. and Finkelstein, A. (2023). Empirical analyses of selection and welfare in insurance markets: a self-indulgent survey. The Geneva Risk and Insurance Review, pages 1–25.
- Fielding, J. and Burningham, K. (2005). Environmental inequality and flood hazard. Local Environment, 10(4):379–395.
- Fielding, J. L. (2012). Inequalities in exposure and awareness of flood risk in England and Wales. *Disasters*, 36(3):477–494.
- Flood Re (2016). Transitioning to an affordable market for household flood insurance: The first flood re transition plan.
- Flood Re (2020). Flood re 2020 annual report.
- Garbarino, N. and Guin, B. (2021). High water, no marks? Biased lending after extreme weather. Journal of Financial Stability, 54:100874.
- Ge, S., Lam, A., and Lewis, R. (2022). The costs of hedging disaster risk and home prices: Evidence from flood insurance. *MIMEO*.
- Genesove, D. and Mayer, C. (2001). Loss aversion and seller behavior: Evidence from the housing market. *The Quarterly Journal of Economics*, 116(4):1233–1260.
- Georgic, W. and Klaiber, H. A. (2022). Stocks, flows, and flood insurance: A nationwide analysis of the capitalized impact of annual premium discounts on housing values. *Journal* of Environmental Economics and Management, 111:102567.
- Giglio, S., Maggiori, M., and Stroebel, J. (2015). Very long-run discount rates. The Quarterly Journal of Economics, 130(1):1–53.
- Gourevitch, J. D., Kousky, C., Liao, Y., Nolte, C., Pollack, A. B., Porter, J. R., and Weill, J. A. (2023). Unpriced climate risk and the potential consequences of overvaluation in us housing markets. *Nature Climate Change*, 13(3):250–257.
- Grainger, C. A. (2012). The distributional effects of pollution regulations: Do renters fully pay for cleaner air? *Journal of Public Economics*, 96(9-10):840–852.

- Hallstrom, D. G. and Smith, V. K. (2005). Market responses to hurricanes. Journal of Environmental Economics and Management, 50(3):541–561.
- Hu, Z. (2022). Social interactions and households' flood insurance decisions. Journal of Financial Economics, 144(2):414–432.
- Imbens, G. W. and Rubin, D. B. (2015). Causal inference in statistics, social, and biomedical sciences. Cambridge university press.
- Imbens, G. W. and Wooldridge, J. M. (2009). Recent developments in the econometrics of program evaluation. *Journal of Economic Literature*, 47(1):5–86.
- Isen, A., Rossin-Slater, M., and Walker, W. R. (2017). Every breath you take—every dollar you'll make: The long-term consequences of the clean air act of 1970. *Journal of Political Economy*, 125(3):848–902.
- Kousky, C. (2010). Learning from extreme events: Risk perceptions after the flood. Land Economics, 86(3):395–422.
- Kousky, C. and Cooke, R. (2012). Explaining the failure to insure catastrophic risks. The Geneva Papers on Risk and Insurance-Issues and Practice, 37:206–227.
- Kuminoff, N. V. and Pope, J. C. (2014). Do "capitalization effects" for public goods reveal the public's willingness to pay? *International Economic Review*, 55(4):1227–1250.
- Lamond, J. and Proverbs, D. (2006). Does the price impact of flooding fade away? *Structural survey*.
- Lustig, H. N. and Van Nieuwerburgh, S. G. (2005). Housing collateral, consumption insurance, and risk premia: An empirical perspective. *The Journal of Finance*, 60(3):1167–1219.
- Mulder, Ρ. (2021).Mismeasuring risk: The welfare effects of flood risk information. URLhttps://faculty. wharton. upenn. edu/wp-content/uploads/2017/07/MismeasuringRisk\_Mulder2021.pdf.
- Murfin, J. and Spiegel, M. (2020). Is the risk of sea level rise capitalized in residential real estate? *The Review of Financial Studies*, 33(3):1217–1255.

- Nguyen, D., Ongena, S., Qi, S., and Sila, V. (2022). Climate change risk and the cost of mortgage credit. *Review of Finance*.
- Niu, D., Eichholtz, P., and Kok, N. (2023). Asymmetric information provision and flood risk salience. MIT Center for Real Estate Research Paper, (23/20).
- Nyce, C., Dumm, R. E., Sirmans, G. S., and Smersh, G. (2015). The capitalization of insurance premiums in house prices. *Journal of Risk and Insurance*, 82(4):891–919.
- Oh, S., Sen, I., and Tenekedjieva, A.-M. (2022). Pricing of climate risk insurance: Regulation and cross-subsidies.
- Payne, R. A. and Abel, G. A. (2012). UK indices of multiple deprivation—a way to make comparisons across constituent countries easier. *Health Stat Q*, 53(22):2015–2016.
- Piazzesi, M., Schneider, M., and Tuzel, S. (2007). Housing, consumption and asset pricing. Journal of Financial Economics, 83(3):531–569.
- Porter, J. and Demeritt, D. (2012). Flood-risk management, mapping, and planning: the institutional politics of decision support in England. *Environment and Planning A*, 44(10):2359–2378.
- Ramcharan, R. (2020). Banks' balance sheets and liquidation values: Evidence from real estate collateral. *The Review of Financial Studies*, 33(2):504–535.
- Ringen, S. (1987). The possibility of politics: a study in the political economy of the welfare state. Oxford University Press.
- Ringen, S. (1988). Direct and indirect measures of poverty. *Journal of Social Policy*, 17(3):351–365.
- Rosen, S. (1974). Hedonic prices and implicit markets: Product differentiation in pure competition. Journal of Political Economy, 82(1):34–55.
- Rözer, V. and Surminski, S. (2021). Current and future flood risk of new build homes across different socio-economic neighbourhoods in england and wales. *Environmental Research Letters*, 16(5):054021.

- Surminski, S. (2018). Fit for purpose and fit for the future? An evaluation of the UK's new flood reinsurance pool. *Risk Management and Insurance Review*, 21(1):33–72.
- Unterberger, C. (2018). How flood damages to public infrastructure affect municipal budget indicators. *Economics of disasters and climate change*, 2(1):5–20.
- Wagner, K. R. (2022). Adaptation and adverse selection in markets for natural disaster insurance. American Economic Journal: Economic Policy, 14(3):380–421.
- Zhang, L. and Leonard, T. (2019). Flood hazards impact on neighborhood house prices. The Journal of Real Estate Finance and Economics, 58(4):656–674.

#### 10 Tables and figures

Variable	Ν	Mean	Std. Dev.	p5	p95
Panel A: Property variables					
ln (Property price)	1,754,067	12.332	0.618	7.313	19.163
D. ln (Property price)	1,754,067	0.424	0.33	-0.019	1.249
New $\text{built}_{t'}$	1,754,067	0.029	0.169	0	1
Property type:					
Detached	1,754,067	0.233	0.423	0	1
Semi-detached	1,754,067	0.288	0.453	0	1
Terraced	1,754,067	0.319	0.466	0	1
Flat	1,754,067	0.153	0.36	0	1
Other	1,754,067	0.008	0.087	0	1
Tenure:					
Freehold	1,754,067	0.801	0.399	0	1
Leasehold	$1,\!754,\!067$	0.199	0.399	0	1
Panel B: Flood risk variables					
Flooded	1,754,067	0.003	0.059	0	1
Flash-flooded	1,754,067	0.001	0.031	0	1
Duration of flood $(100 \text{ days})$	1,754,067	0.001	0.031	0	0
At-risk	1,754,067	0.109	0.312	0	1
Annual flood probability (%)	1,754,067	0.153	0.604	0	0.639
Panel C: Local authority character	istics				
Annual household income	324	42,745.470	$8,\!270.216$	$32,\!338.461$	$57,\!644.445$
Index of Multiple Deprivation	308	19.777	8.012	8.500	34.300
Age	308	42.144	5.094	33.300	50.500
Urban	330	0.727	0.446	0.000	1.000
Education level $(\%)$	324	27.212	7.903	16.900	41.000
Percentage of rental property $(\%)$	300	34.746	9.321	24.475	53.855
Votes for the Green Party $(\%)$	316	2.970	2.007	0.000	5.637
Votes for Brexit (%)	330	54.504	9.963	32.540	68.860

Table 1: Summary statistics

Notes: This table provides descriptive statistics for the variables used in the empirical analysis. Summary statistics of property level variables are presented in Panel A. Panel B summarizes statistics of the measurements of flood risk. Summary statistics of local authorities level variables are shown in Panel C. (ln) denotes that a variable is measured in natural logarithm.

Panel A	Floo	ded	Non-fle	poded	
	Mean	SD	Mean	SD	ND
Property variables					
Detached	0.233	0.423	0.233	0.423	0.000
Flat	0.160	0.367	0.153	0.360	0.014
Other	0.015	0.123	0.008	0.086	0.052
Semi-Detached	0.233	0.423	0.288	0.453	-0.088
Terraced	0.358	0.479	0.319	0.466	0.058
Freehold	0.787	0.410	0.801	0.399	-0.025
Leasehold	0.213	0.410	0.199	0.399	0.025
New	0.035	0.183	0.029	0.169	0.021
Local authority characteristics					
Annual-household-income	43,737.466	8,874.595	42,662.365	7,983.373	0.090
Index-of-multiple-deprivation	17.625	7.552	20.234	8.109	-0.235
Age	42.915	4.915	41.485	5.184	0.200
Urban	0.711	0.453	0.761	0.426	-0.080
Education-level	28.724	6.149	27.303	7.427	0.147
Rental-rate	32.143	5.778	35.052	8.960	-0.273
Vote-for-the-Green-Party	2.721	1.214	3.060	2.310	-0.130
Vote-for-Brexit	52.983	7.319	53.722	9.805	-0.060
Panel B	At-1	risk	No-1	risk	
	Mean	SD	Mean	SD	ND
Property variables					
Detached	0.224	0.417	0.234	0.424	-0.017
Semi-Detached	0.248	0.432	0.293	0.455	-0.072
Terraced	0.338	0.473	0.316	0.465	0.033
Flat	0.182	0.386	0.149	0.356	0.062
Other	0.009	0.092	0.007	0.086	0.009
Freehold	0.786	0.410	0.803	0.398	-0.030
Leasehold	0.214	0.410	0.197	0.398	0.030
New	0.038	0.191	0.028	0.166	0.037
Local authority characteristics					
Annual-household-income	43,243.491	8,745.510	42,594.894	7,885.373	0.055
Index-of-multiple-deprivation	19.726	7.978	20.288	8.123	-0.049
Age	42.123	5.498	41.411	5.138	0.095
Urban	0.689	0.463	0.770	0.421	-0.129
Education-level	27.609	8.831	27.271	7.230	0.030
Rental-rate	35.416	10.342	34.994	8.760	0.031
Vote-for-the-Green-Party	2.852	1.716	3.085	2.369	-0.080
Vote-for-Brexit	54.383	11.327	53.638	9.590	0.050

Table 2: Pairwise test

*Notes:* This table reports statistics of relevant co-variates in the paper. Following Imbens and Wooldridge (2009) and Imbens and Rubin (2015), an absolute normalized difference (ND) smaller than 0.25 indicates no significant difference between the groups.

Panel A	1	2	3	4	5	6
Dependent variable			D. ln (Proj	perty price)		
Flooded	-0.009**	-0.015***	-0.016***			
Flooded v Post Flood Po	(-2.21)	(-2.70)	(-2.97)			
Flooded X 1 0st Flood He		(2.15)	(2.68)			
Flash-flooded	0.002	0.004	0.004			
Flash flooded y Deet Flood Pa	(0.34)	(0.57)	(0.49)			
Fiash-houded x 1 ost Flood He		(0.02)	(0.08)			
At-risk		( )	( )	-0.003***	-0.004***	-0.004***
At rick x Post Flood Ro				(-3.49)	(-3.24)	(-3.11)
At-fisk x 1 ost Flood fie					(3.67)	(3.45)
3 dig plc X Year FE (latest transaction)	Yes	Yes	No	Yes	Yes	No
3 dig plc X Year FE (previous transaction)	Yes	Yes	No	Yes	Yes	No
3 dig plc X Year FE (latest) X Year FE (previous)	No	No	Yes	No	No	Yes
Built year FE	Yes	Yes	Yes	Yes	Yes	Yes
Property controls	No	Yes	Yes	No	Yes	Yes
Observations	1,754,067	1,754,067	$1,\!754,\!067$	$1,\!754,\!067$	1,754,067	1,754,067
R-squared	0.761	0.766	0.792	0.761	0.766	0.792
Panel B	1	2	3	4	5	6
Dependent variable			D. ln (Proj	perty price)		
Flood duration (in 100 days)	-0.015**	-0.023***	-0.024**			
	(-2.55)	(-2.82)	(-2.59)			
Flood duration x Post Flood Re		$0.020^{*}$	$0.023^{**}$			
		(1.90)	(2.04)			
Flood risk mid-point				-0.001	$-0.001^{**}$	-0.001**
				(-1.34)	(-2.50)	(-2.45)
Flood risk mid-point x Post Flood Re					0.001***	0.002***
					(2.82)	(2.61)
3 dig plc X Year FE (latest transaction)	Yes	Yes	No	Yes	Yes	No
3 dig plc X Year FE (previous transaction)	Yes	Yes	No	Yes	Yes	No
3 dig plc X Year FE (latest) X Year FE (previous)	No	No	Yes	No	No	Yes
Built year FE	Yes	Yes	Yes	Yes	Yes	Yes
Property controls	No	Yes	Yes	No	Yes	Yes
Observations	1,754,067	1,754,067	1,754,067	1,754,067	1,754,067	1,754,067
R-squared	0.766	0.766	0.792	0.792	0.766	0.792

#### Table 3: Effect of flood risk and Flood Re on property prices

Notes: The dependent variable in this table is D.(ln) Property price. In Columns 1-3 (4-6) of panel A, the measurements of flood risk in this table is Flooded and Flash-flooded (At-risk). Columns 1 and 4 in panel A present estimation results of equation 1 with the additive fixed effects and without the interaction variable,  $Flooded(At-risk) \times Post Flood Re$ . Columns 2 and 5 of this table present estimation results of equation 1 with the additive fixed effects. Columns 3 and 6 of this table presents estimation results of equation 1. Panel B mirrors panel A with continuous measurements of flood risk, namely Flood duration and Flood risk mid - point. Property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. Standard errors are clustered at local authority district level and the corresponding t-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

Panel A (Extension of the SoP in July 2013)	1	2	
Dependent variable	D. ln (Pro	operty price)	
Flooded	-0.012** -0.014**		
	(-1.97)	(-2.01)	
Flooded x Post SoP extension	0.002	0.003	
	(0.28)	(0.30)	
Flash-flooded	0.001	0.007	
	(0.09)	(0.58)	
Flash-flooded x Post SoP extension	0.006	0.001	
	(0.45)	(0.09)	
3 dig plc X Year FE (latest transaction)	Yes	No	
3 dig plc X Year FE (previous transaction)	Yes	No	
3 dig plc X Year FE (latest) X Year FE (previous)	No	Yes	
Built year FE	Yes	Yes	
Property controls	Yes	Yes	
Observations	933,566	933,566	
$R^2$	0.801	0.822	
Panel B (Monte Carlo simulations of floods and Flood Re)	1 2		
Dependent variable	D. ln (Property price)		
Explanatory variable	Pseudo-flooded	Pseudo-flooded x Post Flood Re	
Rejection rate at the 10% level (2-tailed test)	13.60	11.40	
Rejection rate at the 5% level (2-tailed test)	7.30	7.40	
Rejection rate at the 1% level (2-tailed test)	2.60	1.80	
Coefficient (t-statistics)	-0.002 (-0.50)	$0.003 \ (0.60)$	

Notes: Column 1 in Panel A of this table present estimation results of equation 1 with the additive fixed effects and the placebo treatment, the extension of the SoP. Column 2 in Panel A of this table present estimation results of equation 1 with the placebo treatment *Post SoP extension*. Definitions of variables are detailed in Table A.1 in the Online Appendix. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively. Column 1 (2) of Panel B shows the rejection rates of the null hypothesis of the estimated coefficient of *Pseudo-flooded* (*Pseudo – flooded × Post Flood Re*) being equal to zero at the 10%, 5%, and 1% levels. The last row of panel B presents the corresponding coefficients and *t*-statistics of these two variables.

	1	2	3	4
Dependent variable		D. ln (	Property price)	
Sample	Full s	ample	At least 1 flooded	No flats
Flooded	-0.014*	-0.015***	-0.016***	-0.013**
Flooded x Post Flood Re	(-1.66) $0.018^{***}$	(-2.69) $0.016^{**}$	(-3.08) $0.019^{***}$	(-2.26) $0.019^{***}$
Flash-flooded	(2.63) -0.007	$(2.39) \\ 0.005$	(2.82) 0.004	(2.94) 0.005
Flash-flooded x Post Flood Re	(-0.46) -0.007	(0.70) -0.001	(0.53) 0.001 (0.07)	(0.53) 0.000
At-risk	(-0.51)	(-0.07) -0.003***	(0.07)	(0.02)
At-risk x Post Flood Re		(-2.96) $0.004^{***}$ (3.22)		
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes	Yes
Property controls	Yes	Yes	Yes	Yes
Built year FE X Flooded	Yes	No	No	No
Built year FE X Flash-flooded	Yes	No	No	No
Property controls X Flooded	Yes	No	No	No
Property controls X Flash-flooded	Yes	No	No	No
Observations R-squared	1,754,067 0.792	$1,754,067 \\ 0.792$	57,444 0.790	1,477,458 0.798

Table 5: Robustness test

Notes: This table shows a set of robustness tests. The dependent variable in this table is D.(ln) Property price. Column 1 presents estimation results of equation 1 with additional controls, the interaction of property characteristics with the indicators of flooding events. Column 2 presents estimation results of equation 1 employing both flood events and ex-ante flood risk indicators. Column 3 shows the estimation results with the sample includes only postcodes with at least one flooded property. Column 4 presents estimation results of equation 1 but excluding flats from the sample. Property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. Standard errors are clustered at local authority district level and the corresponding t-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	1	2	က	4	5	9	2	x	6	10	11	12
Dependent variable					D. ln (Prol	perty price)						
Sample split	Ann house inco	ual hold me	Ind mu depri	ltiple ivation	A	9 9	Educ	ation	Urban/ are	Rural	Rental ]	property
	$\geq p50$	< p50	$\geq p50$	< p50	$\geq p50$	< p50	$\geq p50$	< p50	Urban	Rural	$\geq p50$	< p50
Flooded	-0.024***	-0.001	-0.010*	-0.021***	-0.021**	-0.017**	-0.023***	-0.004	-0.022***	-0.003	-0.016**	-0.019**
Flooded x Post Flood Re	(-3.44) $0.019^{**}$	(-0.12) 0.007	(-1.76) 0.004	(-2.99) $0.027^{***}$	(-2.57) $0.029^{***}$	(-2.37) 0.009	(-3.14) $0.023^{**}$	(96.0-)	(-3.54) $0.022^{**}$	(-0.36) 0.008	(-2.56) 0.008	(-2.58) $0.027^{***}$
	(2.38)	(0.70)	(0.43)	(3.46)	(3.29)	(0.78)	(2.47)	(0.88)	(2.57)	(0.91)	(0.75)	(3.00)
Flash-flooded	0.024*	-0.004	-0.001	0.014	0.009	-0.005	0.007	0.005	0.002	0.005	-0.008	0.017
Flash-flooded x Post Flood Re	(1.91)	(64.0-)	(71.0-)	(1.22) 0.002	(0.00) 0.002	(00.0-)	0.000	(0.41) -0.004	(0.14)	(0.010)	(60.0-) -0.006	0.004
	(-0.83)	(0.39)	(0.02)	(0.11)	(0.14)	(-0.07)	(0.01)	(-0.19)	(-0.02)	(0.53)	(-0.28)	(0.20)
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Property controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Chow test <i>F</i> -statistics	2.4	4	2	.71	2.	85	1.	21	3.5	0	5.	64
Observations	878,670	842,721	880,974	790,351	742,261	928, 234	882,770	835,253	1,330,189	413,634	858,243	780,637
$R^{2}$	0.796	0.790	0.788	0.801	0.791	0.795	0.795	0.793	0.793	0.795	0.790	0.796
Power	0.999	0.248	0.249	0.999	0.989	0.826	0.997	0.145	0.995	0.284	0.759	0.999
No. of flooded and flash-flooded properties	3,381	4,204	3,156	4,156	4,230	3,034	4,384	3,115	5,399	2,303	2,991	4,281
Noton: This table succession	lucion motion		tion 1 beg	diff on diffor	and and an	The second			1 (0) in an	lindon and	the second s	and iona
Notes: This table presents estin in local anthority districts with	lation resul hicher (low	ts of equa	tion I bas ce annial	ed on differ income T	rent sub-sa 'he sub-sar	mples. Th nnle in Col	e sub-samț lumn 3 (4)	includes r	mn 1 (2) m monerty tra	cludes pro	perty tran	sactions
in local authority districts with districts with higher (lower) Ind	nigner (lov dex of mult	ver) avera tiple depr	ge annual ivation. T	Income. I The sub-sar	. ne sub-sai nple in Co	npie m Co. dumn 5 (6)	iumn 3 (4) ) includes <sub>1</sub>	nncludes [ property t	property transactions	ansactions s in local .	s 111 local a authority	utnority districts
with higher (lower) age. The su	ib-sample i	n Column	( 7 (8) inc	ludes prop	erty trans	actions in l	ocal autho	rity distri	cts with hig	gher (lowe	r) educati	on level.
The sub-sample in Column 9 ( includes property transactions	10) include in local au	s propert thority di	y transact stricts wit	th higher (	al authorn lower) per	ty districts centage of	rented pro	(rural) art pperties. <sup>7</sup>	ea. The sut The measur	o-sample 1 rements of	n Column f flood risl	11 (12) x in this
table are <i>Flooded</i> and <i>Flash-flo</i>	oded. The	depeñden	t variable	in this tai	ble is $D$ . $h$	$n (Propert_{i})$	$y \ price)$ . F	roperty co	ontrol varis	ubles inclu	ide sets of	dummy
variables muicaung property t detailed in Table A.1 in the Or	ypes, type: uline Annei	ou venur adix. The	е анц мн у Chow te	etuer tue j st. F-statist	property in tic is the <i>j</i>	s new bun 7-statistic	trom a. Ch	ow test fo	ausacuou. r equality (	of the esti	imated coe	efficients
between the two respective sub-	-samples. S	tandard e	strors are	clustered <i>i</i>	at local au	thority dist	trict level <i>i</i>	and the co	rrespondin	g t-statist	ics are rep	orted in
parentheses. *, **, and *** ind	icate statis	tical signi	ficance at	the $10\%$ ,	5%, and 1	% levels, r	espectively		-	0	-	

Table 6. Effect of Flood Re on property prices: Sample splits by demographic characteristics

	1	2	3	4
Dependent variable		D. ln (Prope	rty price)	
Sample split	Percen for the	tage of vote Green Party	Percer vote fo	ntage of r Brexit
	$\geq p50$	< p50	$\geq p50$	< p50
Flooded	-0.016*	-0.017***	-0.007	-0.022***
	(-1.80)	(-2.75)	(-0.84)	(-2.99)
Flooded x Post Flood Re	0.016	0.020**	0.012	$0.022^{**}$
	(1.30)	(2.41)	(1.07)	(2.37)
Flash-flooded	-0.005	0.016	0.004	0.004
	(-0.43)	(1.42)	(0.36)	(0.36)
Flash-flooded x Post Flood Re	0.012	-0.012	-0.002	0.003
	(0.59)	(-0.70)	(-0.10)	(0.22)
Chow test <i>F</i> -statistics		0.50	1	.15
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes	Yes
Property controls	Yes	Yes	Yes	Yes
Observations	889,755	850,770	782,499	961,677
$R^2$	0.798	0.791	0.796	0.791
Power	0.762	0.868	0.227	0.986
No. of flooded and flash-flooded properties	3,162	4,447	3,292	4,374

Table 7: Effect of Flood Re on property prices: Sample splits by revealed beliefs

Notes: This table presents estimation results of equation 1 based on different sub-samples. The sub-sample in Column 1 (2) includes property transactions in local authority districts with higher (lower) percentage of vote for the Green Party. The sub-sample in Column 3 (4) includes property transactions in local authority districts with higher (lower) percentage of vote for Brexit. The measurements of flood risk in this table are *Flooded* and *Flash-flooded*. The dependent variable in this table is *D. ln (Property price)*. Property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. The Chow test *F*-statistic is the *F*-statistic from a Chow test for equality of the estimated coefficients between the two respective sub-samples. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	1	2
Dependent variable	Trade	Days since last trade (ln)
Flooded	-0.036***	0.076***
Flooded x Post Flood Re	(-9.97) $0.060^{***}$	(19.92) -0.131***
Flash-flooded	(9.69) -0.002	(-13.43) $0.070^{***}$
Flash-flooded x Post Flood Re	(-0.25) 0.008 (0.85)	(13.27) -0.115*** (0.26)
3 dig plc X Year FE (latest transaction)	(0.83) Yes	(-9.20) No
3 dig plc X Year FE (latest) X Year FE (previous)	No	Yes
Built year FE	Yes	Yes
Property controls	Yes	Yes
$\frac{\text{Observations}}{R^2}$	$\begin{array}{c} 14,\!446,\!899 \\ 0.014 \end{array}$	1,754,067 0.939

#### Table 8: Effect of flood events and Flood Re on property transactions

Notes: Column 1 of this table presents estimation results of equation 2. The dependent variable in this table is a dummy variable indicates whether the property is traded in the year of observation. Column 2 of this table present estimation results of equation 1 with the dependent variable measuring the natural logarithm of the number of days since the last transaction. The measurements of flood risk in this table are *Flooded* and *Flash-flooded*. Property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	1	2	3	4
Dependent variable	D. ln (Proj	perty price)	Tr	ade
Sample	At-risk	No-risk	At-risk	No-risk
Flooded	-0.025***	-0.001	-0.052***	-0.031***
	(-3.14)	(-0.21)	(-10.91)	(-6.63)
Flooded x Post Flood Re	$0.034^{***}$	0.003	$0.089^{***}$	$0.052^{***}$
	(3.72)	(0.40)	(8.61)	(6.41)
Flash flooded	0.016	0.006	-0.002	0.003
	(1.24)	(0.36)	(-0.22)	(0.38)
Flash flooded x Post Flood Re	0.012	0.006	0.019	0.001
	(0.56)	(0.28)	(1.19)	(0.05)
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes	Yes
Property controls	Yes	Yes	Yes	Yes
Observations	151,624	1,555,018	1,574,118	12,872,223
R-squared	0.829	0.795	0.023	0.014

Table 9: Buyer's attention to flood risk

Notes: Columns 1 and 2 of this table presents estimation results of equation 1 based on different sub-samples. Columns 3 and 4 of this table presents estimation results of equation 2 based on different sub-samples. Sub-samples in Column 1 and 3 (2 and 4) includes property transactions in at-risk (no-risk) areas. The dependent variable in Column 1-2 (3-4) of this table is *D. ln (Property price) (Trade*, a dummy variable indicating the property is transacted in a given year). Property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	1	2	3
Dependent variable	D.	ln (Property pric	ee)
Built year of properties	Up to 2002	From 2002	From 2009
Flooded	-0.016***	0.025	0.020
	(-2.97)	(0.91)	(0.60)
Flooded x Post Flood Re	$0.018^{***}$	-0.009	-0.004
	(2.68)	(-0.25)	(-0.09)
Flash-flooded	0.004	-0.044	0.000
	(0.49)	(-0.80)	(0.00)
Flash-flooded x Post Flood Re	0.001	0.006	0.055
	(0.08)	(0.06)	(0.72)
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes
Property controls	Yes	Yes	Yes
Observations	1,754,067	252,885	137,408
R-squared	0.792	0.585	0.580

Table 10: Effect of flood events and Flood Re on prices by year built

Notes: This table presents estimation results of equation 1 based on different sub-samples. The sub-sample in Column 1 includes property built until 2002. Sub-samples in Column 2 (3) includes property built in 2002 (in 2009) and all following years. Standard errors are clustered at local authority district level and the corresponding t-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

Figure 1: Flood Maps

(a) Flooded postcodes



Notes: These figures show maps of 6-digit postcodes in England that (a) were flooded or (b) were at-risk of flooding in our sample period.





Notes: Each point in the figure represents the estimated coefficient of *Flooded x Post Flood* Re of a specific percentile of the property prices (in the first transaction) distribution and each dash line represents the 95% confidence interval of each estimated coefficient. The specification of the estimations follows equation 1.

#### 11 Online Appendix

#### 11.1 Comparison between Flood Re and National Flood Insurance Program (NFIP)

National Flood Insurance Program (NFIP) in the U.S. shares similar policy objectives of Flood Re in terms of ensuring affordable insurance to households. However, there are subtle differences in several characteristics. The differences of these characteristics also differentiate our work from the two contemporaneous papers, Georgic and Klaiber (2022); Ge et al. (2022), which examine the capitalization effects of flood insurance premiums on property prices under the setting of the NFIP. We summarize the differences between Flood Re and NFIP in Table A.4 in the Online Appendix and discuss the key differences in this section.

**Policy objectives:** Different from Flood Re, which is designed to be a transition plan to a market with risk-based pricing, the NFIP is not designed to phase out in the future and transit the market to risk-based pricing. The NFIP is also responsible for mitigating flood risk across the country and conducts activities beyond providing insurance, such as producing flood maps. In the UK, Flood Re does not play an official role in sharing flood risk information and managing flood risk.

#### Insuree:

The NFIP and Flood Re adopt two different approaches in ensuring the availability of affordable flood insurance. Flood Re offers re-insurance at a discounted price to private insurers, and the lower cost of reinsurance translates to the reduction in insurance premiums of flood insurance. The NFIP directly provides subsidized flood insurance to households, and private insurers serve as distributors of NFIP insurance.

#### Eligibility:

To minimize moral hazard, the NFIP offers insurance only to participating communities which regulate all development in the designated Special Flood Hazard Area (SFHA) in accordance with the NFIP criteria and any applicable State and community floodplain management laws. Furthermore, NFIP insurance is mandatory for households financing properties with federally-backed mortgages. For Flood Re, insurers are free to pick which policies to re-insure through Flood Re and Flood Re insures all policies ceded by insurers, as long as the property was built before 2009.

#### **Funding:**

The main funding source of Flood Re comes for a levee on private insurers, and there is no government support. The NFIP is supported by the government through treasury borrowing in case premiums and annual appropriations for flood-hazard mapping are insufficient to pay its obligations.

#### **Pricing:**

Flood Re's reinsurance premiums are fixed to tax bands. For the NFIP the insurance premiums vary with more factors, e.g., flood zone, design and age of property.

#### Size:

There are around of 4.8 million policies covered by the NFIP, compared to 0.25 million Flood Re policies.

#### 11.2 Conceptual framework

In this section, we consider a simple, one period hedonic pricing model according to which a class of differentiated products is completely described by a vector of measured characteristics (Rosen, 1974). Hence, the price of a property can be characterized by a function of observable property characteristics z, e.g., whether it is a flat or house. It is reduced by the insurance premium which a homeowner pays. This insurance premium is itself a function of flood risk the property is exposed to:<sup>33</sup>

$$Property \, price(z, Premium, Flood \, risk) = f(z) - Premium(Flood \, risk) \tag{3}$$

From equation 3, it can be seen that higher flood risk decreases property price via higher insurance premiums. In mathematical terms, the derivative of property price with respect to flood risk is the negatively proportional to the derivative of insurance premium with respect to flood risk, i.e.  $\frac{\partial Property price}{\partial Flood risk} = -\frac{\partial Premium}{\partial Flood risk}$ .

In absence of a public reinsurance scheme such as Flood Re, insurance companies have a strong incentive to price flood risk into insurance premium, i.e. the derivative of premium with respect to flood risk is positive,  $\frac{\partial Premium}{\partial Floodrisk} > 0$ . As property price is a function of insurance premium, the derivative of property price with respect to flood risk is negative,  $\frac{\partial Property price}{\partial Floodrisk} < 0$ . Hence, we expect to observe higher flood risk to be associated with lower property price.

After the introduction of Flood Re, insurance companies can transfer the flood risk component of their insurance policies to Flood Re. Therefore they have limited incentives to price flood risk into premiums. Thus, we expect the derivative of premium with respect to flood risk to be zero,  $\frac{\partial Premium}{\partial Flood risk} = 0$ . As a result, property price is no longer sensitive to flood risk,  $\frac{\partial Property price}{\partial Flood risk} = 0$ .

<sup>&</sup>lt;sup>33</sup>There is a number of other potential factors affecting insurance premium, such as property structure and claim record. For simplicity, we assume insurance premium is only affected by flood risk of a property.

In our empirical analyses, we examine these conjectures by testing the change in the derivative of property price with respect to flood risk after the introduction of Flood Re, detailed in Section 4 of the paper.

#### 11.3 Dynamic effects of Flood Re on property prices over time

In this section, we implement the following event study regression to leverage time variation:

$$\Delta ln(Price_{i,g,t}) = \beta_0 + \sum_{k=-4,k\neq 0}^4 \beta_k \times Flood \operatorname{Risk}_{i,g,t} \times \mathbb{1}_k^t + \gamma X_{i,g,t} + \delta_{g,t} \times \delta_{g,t'} + \varepsilon_{i,g,t}$$
(4)

where  $\mathbb{1}_{k}^{t}$  is an indicator that reflects the year of a transaction relative to the base year when Flood Re was introduced (so for 2016, k = 0). The coefficients  $\beta_{k}$  represent the effect of flood risk on property prices in each year, both before and after the introduction of Flood Re. We use annual frequency to follow Ge et al. (2022) and to improve the precision of the estimates.

In Figure A.3 (in the Online Appendix), we plot the  $\beta_k$  coefficients from the event study specification (4) using respectively the ex-post and ex-ante proxies for flood risk. Both panels in the figure suggest that flood risk had a negative effect on property prices until the introduction of Flood Re, and no negative effect thereafter. This is in line with our main estimates for aggregated 4-year intervals before and after the introduction of Flood Re in Table 3. While the estimated coefficients are consistently negative in the years before the introduction of Flood Re, some of the estimates are noisy, in particular for 2012 and 2013 when we use flood events as a proxy for flood risk. By estimating year-by-year coefficients we have to rely on a lower number of observations and the number of annual observations depends on the overall number of property transactions and the number of floods which vary significantly over time. For instance, the severe drought in 2010-2012 lead to few number of flooded properties transacted in the following years.<sup>34</sup>

<sup>&</sup>lt;sup>34</sup>Details of 2010-2012 drought, please see MET: 2010-12 drought,

figures
and
tables
Additional
11.4

Table A.1: Variable definitions

Variable	Definition	Source
ln (Property price) D. ln (Property price) New built	Natural logarithm of property price. First difference of the natural logarithm of property price. A dummy variable=1 if the property is newly built in the previous transaction, 0 otherwise.	HM Land Registry Price Paid Data HM Land Registry Price Paid Data HM Land Registry Price Paid Data
Detached Semi-detached Terraced	A dummy variable=1 if the property is a detached house, 0 otherwise. A dummy variable=1 if the property is a semi-detached house, 0 otherwise. A dummy variable=1 if the property is a terraced house. 0 otherwise.	HM Land Registry Price Paid Data HM Land Registry Price Paid Data HM Land Registry Price Paid Data
Flat Other	A dummy variable=1 if the property is a flat, 0 otherwise. A dummy variable=1 if the property is other property type, 0 otherwise.	HM Land Registry Price Paid Data HM Land Registry Price Paid Data
Freehold Leasehold Trade	A dummy variable=1 if the legal ownership of property is freehold, 0 otherwise. A dummy variable=1 if the legal ownership of property is leasehold, 0 otherwise. A dummy variable=1 if the property is transacted in the vear of observation, 0 otherwise.	HM Land Registry Price Paid Data HM Land Registry Price Paid Data HM Land Registry Price Paid Data
Flooded	A dummy variable=1 if the property only experiences flood event last for more than a day four years before the transaction, 0 otherwise.	Environment Agency Recorded Flood Outlines
Flash-flooded	A dummy variable=1 if the property only experiences flood event last for a day four vears before the transaction, 0 otherwise.	Environment Agency Recorded Flood Outlines
Flood duration At-risk Flood risk mid-point Annual household income Index of multiple deprivation Age Education Urban Percentage of rental property Percentage of votes for the Green Party Percentage of votes for the	Duration of flood events in 100 days A dummy variable=1 if the property is in a 6-digit postcode classified as at-risk, 0 otherwise. Expected annual probability of flood. Average annual household income of local authority district in 2019. Index of multiple deprivation of local authority district in 2019. Average age of the households per local authority district in 2019. Proportion of population with level 4 or above qualification, e.g., degree with honours and postgraduate certificate) in local authority district in 2019. A dummy variable=1 if the local authority district in 2019. Percentage of rental residential property of local authority district in 2018 Percentage of rental residential property of local authority district in 2018 Percentage of votes for the Green Party in the 2019 United Kingdom general election of local authority district.	Environment Agency Recorded Flood Outlines Environment Agency Flood Map Environment Agency Flood Map Office of National Statistics Office of National Statistics Office of National Statistics Office of National Statistics Office of National Statistics House of Common Data.gov.uk

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
(1) Annual household income	1								
(2) Index of multiple deprivation	-0.699	1							
(3) Age	-0.02	-0.446	1						
(4) Education level	0.757	-0.514	-0.148	1					
(5) Urban	0.001	0.304	-0.582	-0.007	1				
(6) Percentage of rental property	-0.101	0.599	-0.744	0.224	0.352	1			
(7) Percentage of votes for the Green Party	0.055	-0.077	-0.022	0.189	-0.063	0.111	1		
(8) Percentage of votes for Brexit	-0.534	0.213	0.394	-0.889	-0.129	-0.47	-0.251	1	
(9) (ln) Property price	0.537	-0.391	-0.015	0.489	-0.03	0.023	0.129	-0.394	1

Table A.2: Correlation between local authority variables and property prices

Notes: This table shows the correlation matrix of local authority variables and property prices.

	1	2		
Dependent variable	Awareness of Flood Re			
At-risk	0.154**	-0.340*		
	3.23	-2.14		
At-risk x Age:				
35-54		$0.361^{*}$		
		(2.11)		
>55		$0.455^{**}$		
		(2.89)		
At-risk x Income level:				
26,000-41,599		-0.053		
		(-0.39)		
>41,600		0.163		
		(1.25)		
At-risk x Tax band:				
C-D		0.023		
<b>D H</b>		(0.17)		
E-H		0.019		
		(0.11)		
Age:		0.000		
35-54		-0.303		
		(-1.79)		
>55		-0.275		
T 1 1		(-1.49)		
Income level:		0.110		
26,000-41,599		0.110		
> 41 000		(0.90)		
>41,600		-0.021		
Torr bond		(-0.10)		
C D		0.048		
<u></u> С-D		-0.040		
$\mathbf{F}_{-}\mathbf{H}$		-0.40)		
12-11		-0.091		
		(-0.00)		
Observations	772	455		
<i>R</i> <sup>2</sup>	0.020	0.041		

Table A.3: Awareness of Flood Re

Notes: This table shows the heterogeneity in the awareness of Flood Re among the respondents in the survey of the 2018 Availability and Affordability of Insurance report. The dependent variable in this table is a dummy variable indicating whether the respondent is aware of Flood Re. Standard errors are clustered at region level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	RE	NFIP
Policy objectives 1. Enal	able availability of affordable flood cover for households at flood risk mage a transition to a market with risk-reflective pricing	<ol> <li>Provide access to primary flood insurance</li> <li>Mitigate and reduce the nation's comprehensive flood risk</li> </ol>
Insuree Flood 1	re-insurance is provided to private insurers	Flood insurance is provided to households
Eligibility    1. All r	residential properties built before 2009	<ol> <li>Participating NFIP communities</li> <li>Mandatory for properties with federally-backed mortgages</li> </ol>
Funding [1. Ann 2. Pren	mual Levy from private insurers emium from private insurers	<ol> <li>Premium from households in high flood risk areas</li> <li>Borrowing from the Treasury</li> <li>Annual appropriations</li> </ol>
Pricing   Reinsu	urance premium is fixed to each tax band.	Variable depending on number of factors
Number of covered policies 0.25 mi	nillion	4.8 million

Table A.4: Comparison between Flood Re and NFIP

Notes: This table compares the characteristics of Flood Re and NFIP.

	1	2	c,	4	5	9	2	8	6	10
Dependent variable					D. ln (Pr	operty price				
Sample split	Ann	ual	Inc	lex of		Age	Educ	ation	Urban/	Rural
•	house inco	hold me	mu depr	ltiple ivation		)			are	ಹ
	$\geq p50$	< p50	$\geq p50$	< p50	$\geq p50$	< p50	$\geq p50$	< p50	Urban	Rural
At-risk	-0.005***	-0.001	-0.003	-0.005***	-0.002	-0.006***	-0.006***	-0.001	-0.006***	0.000
	(-3.50)	(-0.54)	(-1.40)	(-3.01)	(-1.17)	(-3.52)	(-3.46)	(-0.66)	(-4.14)	(0.08)
At-risk x Post Flood Re	$0.007^{***}$	(0.47)	(1.46)	$0.006^{***}$	(1.37)	$0.007^{***}$	$0.007^{***}$	(0.62)	$0.006^{***}$	(0.39)
2 discriberty V Vran Effe (1940-4) V Vran Effe (manifered)	(1011)	Vac.	Var.	Vac.	(10.1)	(toro)	Var.	Vac	V <sub>22</sub>	Vacua)
3 dig pic A rear FE (latest) A rear FE (previous)	Ies	res	res	res	res	Ies	res	res	Ies	res
Built year FE	$\mathbf{Yes}$	Yes	Yes	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	Yes
Property controls	Yes	Yes	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes	Yes	Yes	Yes
Chow test <i>F</i> -statistics	2.4	11	с.)	3.24	7	1.49	2.	29	6.6	2
Observations $R^2$	$878,670 \\ 0.796$	842,721 0.790	$880,974 \\ 0.788$	790,351 0.801	$742,261 \\ 0.791$	$928,234 \\ 0.795$	$882,770 \\ 0.795$	835,253 0.793	$1,330,189\\0.793$	$413,634 \\ 0.795$
Notes: This table presents estimation results of economical authority districts with higher (lower) average with higher (lower) Index of multiple deprivation. age. Sample in Column 7 (8) includes property transactions in local authority variable in this table is D. In (Property price) and and whether the property is new built in the previous test $F$ -statistic is the $F$ -statistic from a Chow test clustered at local authority district level and the other the property district level and the construction.	equation 1 h ge annual i Sample in ransactions districts in l property c ious transac for equality correspondii	ased on $c$ ased on $c$ column $5$ in local $i$ urban ( $u$ ontrol va tion. Def the $c$ r of the $cr$	different s ample in (6) inclu authority ural) area riables inc initions o stimated o stimated o	sub-sample column 3 des proper districts w a. Measur clude sets ( f variables coefficients eported in	s. Sample (4) incluc (4) incluc rith transaa rith highe ements of ements of dummy are detail between parenthee	in Colum les propert, stions in loo r (lower) ed ' flood risk variables i the two res ses. *, **, *	n 1 (2) inc y transaction cal authorit ducation le in this tal ndicating I e A.1 in the pective sul-	ludes proj ons in loc: y districts vel. Samp or el. At-1 oroperty t, c Online A -samples.	perty trans: al authority s with high ole in Colur isk. The d ypes, types ypes, types typendix. 7 Standard o istical signi	ctions in districts r (lower) m 9 (10) ependent of tenure he Chow rrrors are icance at

رامنن -Ч . 4+ 4+:-4 ÷ --1:+;[ -Ŭ • à f El + БÆ ь. Ц < Table

	1	2	3	4	
Dependent variable		D. ln (Property price)			
Sample split	Percenta for the G	ge of vote reen Party	Percer vote fo	ntage of r Brexit	
	$\geq p50$	< p50	$\geq p50$	< p50	
At-risk	-0.004**	-0.004***	-0.002	-0.006***	
	(-2.13)	(-2.64)	(-0.93)	(-3.38)	
At-risk x Post Flood Re	$0.006^{***}$	$0.003^{**}$	0.003	$0.006^{***}$	
	(2.91)	(2.14)	(1.43)	(3.60)	
Chow test <i>F</i> -statistics	0.	37	2	.34	
3 dig plc X Year FE (latest) x Year FE (previous)	Yes	Yes	Yes	Yes	
Built year FE	Yes	Yes	Yes	Yes	
Property controls	Yes	Yes	Yes	Yes	
Observations	889,755	850,770	782,499	961,677	
$R^2$	0.798	0.791	0.796	0.791	

Table A.6: Effect of Flood Re on property price: Sample split by revealed beliefs with the *ex-ante* flood risk measurement

Notes: This table presents estimation results of equation 1 based on different sub-samples. Sample in Column 1 (2) includes property transactions in local authority districts with higher (lower) percentage of vote for the Green Party. Sample in Column 3 (4) includes property transactions in local authority districts with higher (lower) percentage of vote for Brexit. Measurements of flood risk in this table is At-risk. The dependent variable in this table is D. In (Property price) and property control variables include sets of dummy variables indicating property types, types of tenure and whether the property is new built in the previous transaction. Definitions of variables are detailed in Table A.1 in the Online Appendix. The Chow test *F*-statistic is the *F*-statistic from a Chow test for equality of the estimated coefficients between the two respective sub-samples. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

	1	2	3
Dependent variable	D.	ln (Property pric	ee)
Built year of properties	Up to 2002	From 2002	From 2009
At-risk	-0.004***	0.002	-0.002
	(-3.11)	(0.51)	(-0.31)
At-risk x Post Flood Re	$0.004^{***}$	0.007	0.012
	(3.45)	(1.33)	(1.46)
3 dig plc X Year FE (latest) X Year FE (previous)	Yes	Yes	Yes
Built year FE	Yes	Yes	Yes
Property controls	Yes	Yes	Yes
Observations	1,754,067	252,885	137,408
R-squared	0.792	0.585	0.580

Table A.7: Effect of flood events and Flood Re on prices of recently-built properties with the *ex-ante* flood risk measurement

Notes: This table presents estimation results of equation 1 based on different sub-samples. The sub-sample in Column 1 includes properties built until 2002. Sub-sample in Column 2 (3) includes properties built in 2002 (in 2009) and all following years. Standard errors are clustered at local authority district level and the corresponding *t*-statistics are reported in parentheses. \*, \*\*, and \*\*\* indicate statistical significance at the 10%, 5%, and 1% levels, respectively.

#### Figure A.1: Flood Re mechanism



Notes: This figure was produced in Crick et al. (2018), depicting the mechanism of Flood Re and the interplay between different key players of Flood Re.



Figure A.2: Dynamic effects of flood on property prices before the introduction of Flood Re

Notes: The figure reports the estimated  $\beta_k$  of the following equation:  $\Delta ln(Price_{i,g,t}) = \beta_0 + \sum_{k=1}^6 \beta_k Flooded_{i,g,t} \times \mathbb{1}_k^t + \gamma X_{i,g,t} + \delta_{g,t} \times \delta_{g,t'} + \varepsilon_{i,g,t}$ .  $\mathbb{1}_k^t$  is an indicator that takes a value of one if k is in t number of year after flooding. Standard errors are clustered at local authority district level. The sample of the estimation only include observations before the introduction of Flood Re.



Figure A.3: Dynamic effects of Flood Re on property prices over time

Notes: Panel A of the figure reports the estimated  $\beta_k$  of the following equation:  $\Delta ln(Price_{i,g,t}) = \beta_0 + \sum_{k=-4,k\neq 0}^{4} \beta_k Flooded_{i,g,t} \times \mathbb{1}_k^t + \gamma X_{i,g,t} + \delta_{g,t} \times \delta_{g,t'} + \varepsilon_{i,g,t}$ . Panel B reports the estimated  $\beta_k$  of the following equation:  $\Delta ln(Price_{i,g,t}) = \beta_0 + \sum_{k=-4,k\neq 0}^{4} \beta_k At - risk_{i,g} \times \mathbb{1}_k^t + \gamma X_{i,g,t} + \delta_{g,t} \times \delta_{g,t'} + \varepsilon_{i,g,t}$ .  $\mathbb{1}_k^t$  is an indicator that takes a value of one if k is in year t.